If you have issues viewing or accessing this file contact us at NCJRS.gov.

National Criminal Justice Reference Service



This microfiche was produced from documents received for inclusion in the NCJRS data base. Since NCJRS cannot exercise control over the physical condition of the documents submitted, the individual frame quality will vary. The resolution chart on this frame may be used to evaluate the document quality.



MICROCOPY RESOLUTION TEST CHART NATIONAL RUPEAU OF STANDARDS-1963-

Microfilming procedures used to create this fiche comply with the standards set forth in 41CFR 101-11.504.

Points of view or opinions stated in this document are those of the author(s) and do not represent the official position or policies of the U.S. Department of Justice.

National Institute of Justice United States Department of Justice Washington, D.C. 20531

DATE FILMED

11/9/81

U.S. Department of Justice Bureau of Justice Statistics

Issues in the Measurement of Victimization

The task at hand

Conceptual issues





asurement issues

Bureau of Justice Statistics Reports Applications of the National Crime

Single copies are available at no charge from the National Criminal Justice Reference Service, Box 6000, Rockville, Md, 20850, Multiple copies are for sale by the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402.

National Crime Survey:

- Criminal Victimization in the United States (annual):
- Summary Findings of 1978-79 Changes in Crime and of Trends Since 1973, NCJ-62993
- A Description of Trends from 1973 to 1978, NCJ-66716
- 1978 (final report), NCJ-66480
- 1977, NCJ-58725
- 1976, NCJ-49543
- 1975, NCJ-44593
- 1974, NCJ-39467
- *1973. NCJ-34732

The Cost of Negligence: Losses from Preventable Household Burglaries, NCJ-53527

The Hispanic Victim: Advance Report, NCJ-67706

Intimate Victims: A Study of Violence Among Friends and Relatives, NCJ-62319 Crime and Seasonality, NCJ-64818

Criminal Victimization of New York State Residents, 1974-77, NCJ-66481

Criminal Victimization of California Residents, 1974-77, NCJ-70944

Indicators of Crime and Criminal Justice: Quantitative Studies, NCJ-62349 **Criminal Victimization Surveys in 13** American cities (summary report, 1 vol.). NCJ-18471 Boston, NCJ-34818 Buffalo, NCJ-34820 Cincinnati, NCJ-34819 Houston, NCJ-34821 Miami, NCJ-34822 Milwaukee, NCJ-34823 Minneapolis, NCJ-34824 New Orleans, NCJ-34825 Oakland, NCJ-34826

Pittsburgh, NCJ-34827 San Diego, NCJ-34828 San Francisco, NCJ-34829 * Washington, D.C., NCJ-34830 Public Attitudes About Crime (13 vols.) Boston, NCJ-46235 Buffalo, NCJ-46236 Cincinnati, NCJ-46237 Houston, NCJ-46238 Miami, NCJ-46239

Milwaukee, NCJ-46240 * Minneapolis, NCJ-46241 New Orleans, NCJ-46242 Oakland, NCJ-46243 Pittsburgh, NCJ-46244 San Diego, NCJ-46245 San Francisco, NCJ-46246

- Washington, D.C., NCJ-46247 * Criminal Victimization Surveys in Chicago, Detroit, Los Angeles, New York, and Philadelphia: A Comparison of 1972 and 1974 Findings, NCJ-36360
- **Criminal Victimization Surveys in Eight** American Cities: A Comparison of 1971/72 and 1974/75 Findings—National Crime Surveys in Atlanta, Baltimore, Cleveland, Dallas, Denver, Newark, Portland, and St. Louis, NCJ-36361
- * Criminal Victimization Surveys in the Nation's Five Largest Cities: National Crime Panel Surveys in Chicago, Detroit. Los Angeles, New York, and Philadelphia, 1972, NCJ-16909
- * Crimes and Victims: A Report on the Dayton/ San Jose Pilot Survey of Victimization NCJ-013314

*out of stock but available on interlibrary loan

Survey Victimization and Attitude Data: Public Opinion About Crime: The Attitudes of Victims and Nonvictims in Selected

Cities, NCJ-41336 Local Victim Surveys: A Review of the

- Issues, NCJ-39973 *The Police and Public Opinion: An Analysis of Victimization and Attitude Data from
- 13 American Cities, NCJ-42018 An Introduction to the National Crime
- Survey, NCJ-43732 **Compensating Victims of Violent Crime:**
- Potential Costs and Coverage of a National Program, NCJ-43387 Rape Victimization in 26 American Cities,
- NCJ-55878 Crime Against Persons in Urban, Suburban,
- and Rural Areas: A Comparative Analysis of Victimization Rates, NCJ-53551
- Criminal Victimization in Urban Schools, NCJ-56396
- Restitution to Victims of Personal and Household Crimes, NCJ-72770
- Myths and Realities About Crime: A Nontechnical Presentation of Selected Information from the National Prisoner Statistics Program and the National Crime Survey, NCJ-46249

National Prisoner Statistics: Capital Punishment (annual)

1979, NCJ-70945 Prisoners in State and Federal Institutions on December 31:

- 1979. NCJ-73719 *Census of State Correctional Facilities, 1974
- advance report, NCJ-25642 **Profile of State Prison Inmates:** Sociodemographic Findings from the 1974 Survey of Inmates of State Correctional
- Facilities, NCJ-58257 *Census of Prisoners in State Correctional Facilities, 1973, NCJ-34729
- Census of Jails and Survey of Jail Inmates, 1978, preliminary report, NCJ-55172 Profile of Inmates of Local Jails: Sociodemographic Findings from the 1978 Survey
- of Inmates of Local Jails, NCJ-65412 *The Nation's Jails: A report on the census of jails from the 1972 Survey of Inmates of Local Jails, NCJ-19067

Uniform Parole Reports:

- Parole in the United States (annual) 1979, NCJ-69562
- 1978, NCJ-58722 1976 and 1977, NCJ-49702
- A National Survey of Parole-Related Legislation Enacted During the 1979 Legislative Session, NCJ-64218

Characteristics of the Parole Population, 1978, NCJ-66479

- Children in Custody: Juvenile Detention and **Correctional Facility Census**
- 1977 advance report: Census of Public Juvenile Facilities, NCJ-60967 Census of Private Juvenile Facilities, NCJ-60968
- 1975 (final report), NCJ-58139 1974, NCJ-57946 1973, NCJ-44777 *1971. NCJ-13403

- State and Local Probation and Parole Systems, NCJ-41335
- State and Local Prosecution and Civil Attorney Systems, NCJ-41334
- National Survey of Court Organization: 1977 Supplement to State Judicial Systems, NCJ-40022
- *1975 Supplement to State Judicial Systems, NCJ-29433
- 1971 (full report), NCJ-11427 State Court Model Statistical Dictionary, NCJ-62320
- State Court Caseload Statistics: The State of the Art, NCJ-46934
- Annual Report, 1975, NCJ-51885 Annual Report, 1976, NCJ-56599
- A Cross-City Comparison of Felony Case Processing, NCJ-55171
- Trends in Expenditure and Employment Data for the Criminal Justice System, 1971-77 (annual), NCJ-57463
- Expenditure and Employment Data for the Criminal Justice System (annual) 1979 advance report, NCJ-73288 1978 Summary Report, NCJ-66483
- 1978 final report, NCJ-66482 1977 final report, NCJ-53206
- Justice Agencies in the U.S.: Summary Report of the National Justice Agency List, NCJ-65560
- Dictionary of Criminal Justice Data Terminology: Terms and Definitions Proposed for Interstate and National Data Collection and Exchange, NCJ-36747

Utilization of Criminal Justice Statistics Project:

- Sourcebcok of Criminal Justice Statistics 1980 (annual), NCJ-71096
- * Offender-Based Transaction Statistics: New Directions in Data Collection and Reporting, NCJ-29645
- Sentencing of California Felony Offenders, NCJ-29646
- Crime-Specific Analysis: * The Characteristics of Burglary In Ments, NCJ-42093
- An Empirical Examination of Burgiury Offender Characteristics, NCJ-43131
- *An Empirical Examination of Burglary Offenders and Offense Characteristics, NCJ-42476
- Federal Criminal Sentencing: Perspectives of Analysis and a Design for Research, NCJ-33683
- Variations in Federal Criminal Sentences: A Statistical Assessment at the National Level, NCJ-33684
- Federal Sentencing Patterns: A Study of
- Geographical Variations, NCJ-33685 Predicting Sentences in Federal Courts: The Feasibility of a National Sentencing Policy, NCJ-33686

1

Sources of National Criminal Justice Statistics: An Annotated Bibliography, NCJ-45006

U.S. Department of Justice Bureau of Justice Statistics



Issues in the Measurement of Victimization

NCJ-74682, June 1981

by Wesley G. Skogan Northwestern University Evanston, Illinois 60201

74682

U.S. Department of Justice National Institute of Justice

This document has been reproduced exactly as received from the person or organization originating it. Points of view or opinions stated in this document are those of the authors and do not necessarily represent the official position or policies of the National Institute of Justice.

Permission to reproduce this consisted material has been dranted by

	_
Bureau of Justice	

to the National Criminal Justice Reference Service (NCJRS).

Further reproduction outside of the NCJRS system requires permission of the convictor owner

U.S. Department of Justice Bureau of Justice Statistics

Benjamin H. Renshaw, III Acting Director

Charles R. Kindermann, Ph.D. Acting Director Statistics Division

This project was supported by Grant No. 78-SS-AX-0045, awarded to the Center for Urban Affairs, Northwestern University by the Statistics Division, National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration (now Bureau of Justice Statistics), U.S. Department of Justice, under the Omnibus Crime Control and Safe Streets Act of 1968, as amended. The project was monitored by Patsy A. Klaus of the Bureau of Justice Statistics. Preparation of the manuscript was supervised by Marlene B. Simon. She and Paul J. Lavrakas, along with several anonymous reviewers, contributed helpful comments on an earlier version of the report. Points of view or opinions stated in this document are those of the author(s) and do not necessarily represent the official position or policies of the U.S. Department of Justice.

BJS authorizes any person to reproduce, publish, translate, or otherwise use all or any part of the material in this publication, with the exception of those items indicating that they are copyrighted by or reprinted by permission from any other source.

For sale by the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402

ii

Abstract

This volume summarizes 15 years of research on methodological issues in the measurement of criminal victimization by means of population surveys. The report reviews some features of crime which affect our ability to measure it accurately, including the relative infrequency of serious victimization, the skewed distribution of victimization in the population, and the furtive character of crime. The third chapter addresses issues related to the operationalization of victimization in survey questionnaires. It examines the events orientation of victimization surveys, the assumption that crimes always are discrete incidents rather than continuous social processes, and the utility of measures of criminal activity abstracted from their social context. The fourth chapter reviews specific measurement problems: limited distribution of knowledge of incidents, forgetting or inaccurate recall of events, and differential productivity of survey respondents. The next chapter reviews three procedural issues which affect estimates of victimization rates: problems of panel bias and attrition, differences between telephone and in-person interviews, and interviewer effects. The final chapter 5 summarizes the current state of the art in this area and discusses possible future developments in victimization survey methodology.

Contents

Abstract, ii

Chapters

1. Introduction, 1 Methodological features, 1

2. The task at hand, 2 Frequency of crime, 2 Distribution of crime, 4 Furtive nature of crime, 5

3. Conceptual issues, 7 Operationalization of victimization, 7 Discrete events versus continuous processes, 7 Events orientation, 9 Social meaning of incidents, 9

4. Measurement issues, 11 Methodological research techniques, 11 Analytic methods, 11 Experiments, 12 Record checks, 13 Knowledge of incidents, 14 Forgetting and not telling of incidents, 15 Load and fatigue, 15 Lying and not telling, 16 Forgetting, 17 Conflicting evidence, 18 Inaccurate or incomplete recall of incidents, 19 Telescoping, 19 Other sources of measurement error, 21 Differential productivity of respondents, 22 Summary, 22

5. Procedural issues, 25 Panel bias and attrition, 25 Panel bias, 25 Panel attrition, 26 Telephone versus personal interviews, 26 Research results, 27 Interviewer effects, 27

6. Assessment, 29 Current state of the art, 29 Interpersonal violence, 30 Future developments, 31 Series incidents, 31

References, 3.3

Figures

1. Survey estimates of city crime rates and their confidence intervals, 3

2. Individual screen questions, δ

3. Rates of victimization reported, by month of recall, 17

4. Source of method variance in victimization data, 23

Tables

1. Assault and robbery victimization, by number of times victimized, 4

2. Distribution of "don't know" responses about crime, by type of crime, 5

3. Rate of assaultive violence, by educational attainment, I0

4. Patterns of record-check non-response, 16

5. Record-check recall, by months of recall demanded, 17

6. Victimization rate estimates based on bounded and unbounded national samples, 20

Chapter 1

Introduction

This volume presents an extensive overview of 15 years of methodological development in refining the methods by which criminal victimization can be measured through survey interviews. Both in the United States and abroad, there has been a great deal of research on methods for employing sample surveys to gauge the incidence of crime. This report attempts to synthesize those efforts, to report on the nature of the problems facing those conducting victimization surveys, and to assess the current state of the art. It examines the conceptualization of victimization implicit in current survey methods, the measurement techniques employed in interviews, and the procedures utilized in conducting surveys to gather data on the crime experiences of the public. The review takes a critical stance with regard to many of those concepts and methods. However, this detailed criticism is possible only because those responsible for the surveys have paid a great deal of attention to methodological aspects of their work. The Bureau of Justice Statistics (BJS)* and the U.S. Census Bureau have sponsored a number of investigations of victim survey methodology, and they continue to test the limitations of the data which their current survey produces. Reports of those efforts form the basis for much of this review. This volume also attempts to place in context these methodological problems and the efforts to solve them by reviewing related surveys and the results of more general methodological investigations. That comparative analysis suggests that many of the problems facing victimization researchers are not unique to their domain and that their solutions for those problems are generally as effective as most.

This volume is aimed at interested and relatively well-informed crime researchers and users of victimization data. It is

*The National Crime Survey was originally developed within the Law Enforcement Assistance Administration and was transferred to the newly created Bureau of Justice Statistics in 1980.

not a "how to do it" text, but a synthesis of a substantial body of methodological research on the reliability and validity of victimization data. It is intended to inform researchers and data users of the strengths and weaknesses of those data, and, where weaknesses are apparent, to pinpoint some specific pitfalls to avoid in using the data. This report also notes a number of areas in which further methodological research is needed. There is increasing interest in criminal justice statistics with national scope, and attention to these continuing problems may speed the development of a useful Federal statistical program.

Methodological features

The specific focus of this volume is the methodological features of the National Crime Survey. This survey program, sponsored by BJS and conducted by the U.S. Census Bureau, is our primary source of data on patterns of criminal victimization, and the only ongoing victimization survey which is national in scope. (For details on the National Crime Survey see: Garofalo and Hindelang, 1977.) However, there has been a great deal of innovative methodological research conducted by States and municipalities, and abroad, and many of these contributions will also be reviewed here. The findings of all of those investigations have been remarkably consistent, suggesting that the problems inherent in measuring the extent of criminal victimization through sample surveys involve fundamental social and psychological issues.

Some of the consistency may reflect the fact that all of the surveys have focused on the same issue-victimization. The next chapter examines some aspects of crime itself and how they shape the nature of the survey enterprise. These include the relative infrequency of serious victimization, the skewed distribution of victimization in the population, and the covert nature of criminal action. Chapter 3 explores certain conceptual issues related to how surveys operationalize the concept of criminal victimization. It criticizes the events orientation of such surveys, the characterization of crime as a discrete rather than an occasionally continuous social process, and the abstraction of events from their social context. Chapter 4 reviews the impact of human factors on the accurate recall of events from the past. These contribute to the selective forgetting or inaccurate recall of those incidents. Chapter 5 examines problems which are procedural in character, including panel bias and attrition, interviewer effects, and decisions regarding interviewing procedures. Chapter 6 concludes with an assessment of current survey practices, a critique of data on the incidence of criminal assault, and some notes on future methodological developments in this area.

The task at hand

Many methodological and procedural barriers to measuring victimization can be overcome. However, attempts to measure the frequency of crime through general population surveys face certain ineluctable difficulties stemming from the nature of crime. These difficulties ultimately limit our ability to use survey methods in gathering data on offenses. Three facts about crime are important in this regard: It is relatively infrequent, especially in its most serious and violent forms; it is unevenly distributed; and most criminals do their best to avoid detection.

Frequency of crime

Despite the large numbers which fill the columns of the FBI's Uniform Crime Report, one of the most important aspects of crime from the point of view of individual citizens is that it strikes infrequently. In any reasonably brief time period most people are not victimized. Further, there is a generally inverse relationship between the seriousness of a crime and its frequency, Recently, Marvin Wolfgang (1978) and his associates developed a supplemental questionnaire for the National Crime Survey which probed the perceived seriousness of many kinds of offenses in the eyes of the general population. The most frequent crimes-those which strike more than once for every 100 persons in the United States-all scored in the lower ranges of seriousness. This category included such offenses as petty theft (which scored 1.7 on their seriousness scale) and trivial violence (1.5). Some less serious offenses also are uncommon (like pocket picking), but no truly serious crime is very frequent. Those nearer the top of the seriousness scale, such as forcible rape (25.8) and homicide (35.6), all have incidence rates of less than 1 in 10,000.

The infrequency of serious crime in the general population has important consequences for victimization surveys. This was a major concern when the Crime Commission first considered conducting a victimization survey. Official statistics for the mid-1960's suggested, for example,

2

that there were about 180 robberies of all kinds (including crime against businesses) for every 100,000 persons in the population. Given the apparently low frequency of such incidents, very large survey samples would be required to collect reliable information on many kinds of serious crime. All surveys are subject to sampling error, and, if victimization rates are relatively low, sampling error may lead to very substantial variations in estimates of those rates. Sampling error may also make it difficult to examine differences in victimization among population groups, because sampling variation may be greater than most true differences among them. Those who favored the surveys were convinced that the "dark figure" of unreported crime was large enough that a national sample of 10,000 households would uncover enough victims for analysis. The truth lay somewhere in between. While the survey collected data on property crimes which were common enough to make estimates of their incidence in the population, the number of victims of personal crimes who were located was

The problem is more extreme in countries with lower crime rates. There random sample surveys of reasonable (and affordable) size will not allow reliable inferences either about rates of victimization or the characteristics of those affected by crime. For example, two surveys of people in Scandinavian countries (almost 2,000 people in Denmark and 1,500 in Norway) uncovered respectively 84 and 45 self-reported victims of violence or threats during the previous 2 years (Wolf, 1976b). In Göttingen, Germany, Schwind et al. (1975) found only 49 robbery victims in a sample of 1,170 persons. Sampling variances associated with national estimates of victimization rates made from data distributed in this fashion are extraordinarily large. It is unlikely

very small.

that reliable differences between the personal crime victimization rates of any two European countries can be documented.

Even with its high victimization rate, the United States presents similar difficulties. Because crimes are so infrequent, researchers have to look long and hard to find recent victims of many of them. Most surveys involve relatively few people. A survey of the United States normally requires a sample of only about 1,500 carefully chosen respondents to gather reliable information about the personal attributes of people and things that are on their minds-their incomes, consumption habits, and attitudes. Virtually everyone interviewed has something to report on all of these topics, and differences among groups often are much larger than sampling error associated with samples of that size. In contrast, victimization surveys use samples of people to gather reports of particular kinds of events. Most of those who are interviewed have little information to offer, and it therefore is necessary to interview many of them. This is the major reason why LEAA's survey program has employed such large samples: 60,000 households for the national panel and 10,000 households in the city surveys. However, even based on 10,000 households (and 2,500 commercial establishments), the city surveys conducted by the U.S. Census Bureau for LEAA still could not produce estimates of victimization rates which were large relative to the magnitude of sampling error. It can be difficult to make reliable judgments about differences between and among cities due to the sampling error clouding those estimates. The surveys did not uncover many victims, especially of personal crimes. For example, estimates for rape victimization in Philadelphia were based on 29 actual interviews with rape victims, and in Detroit only about 150 robbery victims were encountered in the household survey (Jacob, 1975). While the data resulting from the city surveys have been used to make estimates of victimization rates for each city, those rates may vary enormously within the confidence interval which surrounds them. Figure 1 illustrates the problem.



Figure 1 presents victimization rate estimates and their standard errors for eight cities where interviews were conducted in 1972. It reports estimates of the commercial burglary rate and the personal

crime victimization rate for each of the cities, and the range within which we are 95 percent certain the "true" value must fall. The size of these intervals represents only the potential effect of sampling error.

Cities such as Newark, Baltimore, St. Louis, and Denver have virtually indistinguishable commercial burglary victimization rates despite large differences in their estimated values (63 per 1,000 in Newark, 44 per 1,000 in Denver), because the confidence intervals around these figures largely overlap.

The situation is less problematic when we examine personal victimizations, because the samples used to gather these data were larger than those for the commercial surveys. While it is difficult to talk about the difference between the personal crime rates of any two adjacent cities represented in figure 1, across the group of eight cities some are clearly high-crime and some low-crime places.

These confidence intervals do not take into account other, nonsampling errors. As we shall see, Bailey et al. (1978) recommend that confidence intervals should be doubled to protect against errors in inference due to the very considerable effect of interinterviewer differences on victimization reports. Because of the frequency-of-events problem and the many other measurement issues involved here, it is extremely difficult to produce accurate estimates of the victimization rate even when large samples are brought to bear on the subject.

The relative infrequency of victimization by serious crime has significant implications for the utility of the data generated by the crime surveys and for the applicability of the method in criminal justice. research. The National Crime Survey originally was designed to produce estimates of the crime rate on a quarterly basis. This, in large measure, shaped the methodological development of the survey, for it focused attention on the ability of victims to recall the correct dates of incidents so that crimes could be counted in the appropriate quarter. There was less attention given to other possible criteria, including the accuracy with which other aspects of crimes could be remembered. However, quarterly estimates are plagued by a substantial amount of sampling error. and to date the data have only been used to make yearly estimates of victimization rates. The substantial sampling errors associated with estimates of victimization rates for cities limit their utility for city-

3

The relative infrequency of victimization by serious crime has significant implications for the utility of the data generated by the crime surveys and for the applicability of the method in criminal justice research.

level analyses as well. Some researchers (cf. Booth et al., 1977; Shichor et al., 1979) have used published estimates of rates for cities as indicators in aggregatelevel correlational studies of patterns of victimization; however, none has taken into account the extent of, and variation in, error in those estimates.

Finally, the large sampling error associated with estimates of rates of victimization made from reasonably large samples greatly limits the utility of the technique for individual researchers or even local governments. For example, surveys conducted to evaluate the Seattle Community Crime Prevention Program produced "before and after" victimization rates for experimental areas which indicated a 36 percent reduction in burglary-but that difference was not large enough to be statistically significant. This was true in spite of the fact that (1) almost 1,500 households were questioned in the first round of interviewing and 1,200 in the second; (2) these interviews all were concentrated in only five census tracts; and (3) the survey examined the incidence of the most frequent major crime (Cirel et al., 1977). Studies that focus on personal crimes or small subgroups in the population are limited to an even greater extent by the laws of sampling.

Distribution of crime

Not everyone shares equally the burden of crime, and the highest risk groups often are relatively small. They also can be the most difficult to locate. One goal of victimization surveys is to identify high-risk subpopulations and their particular problems. Because these groups contribute disproportionately to the overall victimization rate, the payoff from these data for law enforcement officials could be substantial. Researchers interested in the effects of crime on fear and behavior also could benefit from detailed data on such groups as minority-group males, senior citizens, and middle-class families moving back into America's central cities. Because these groups all are relatively small, general samples of the population will not yield sufficient numbers for analysis. Further, ordinary survey techniques often fail to represent everyone in the population proportionately. The U.S. Population Census of 1970, for example, appears to have undercounted black males aged 30 to

Table 1. Assault and robben by number of times	y victimization, victimized			· · · · · ·	
Number of times victimized in past 6 months	Ass	ault	Rob	bery	
0	134,713	98.8%	135,943	99.7%	
1	1,327	1.0%	384	0.3%	
2	141	0.1%	16	*	
3-4	98	-	10		
5-6	44	,	7		
7-12	34	· '	4	· · · · · ·	
13 or more	8	- -	1		· · ·
	136,365	100%	136,365	100%	

* "---" indicates percentage less than 0.1%. Source: Author's computation. Note that this data was weighted to reflect noninterviews, sampling considerations, and other aspects of NCS weighting not related to producing population estimates. The frequency of series incidents was estimated using the mid-points of their frequency categorization.

34 by 18 percent, and the National Crime Survey for 1974 reached only 68 percent of the targeted figure for that group (National Research Council, 1976:table 6).

While crime is relatively infrequent in the general population, this is not the case among certain subgroups. Crime is spatially segregated. For example, in 1970 two-thirds of the reported robberies in the United States were concentrated in 32 cities which housed only 16 percent of the nation's population (Skogan, 1979). Within those cities crime was also heavily concentrated in a few places. In general, differences in the distribution of crime within cities are even greater than differences among cities. As a result, a relatively small proportion of the population is exposed to extremely high levels of risk, especially from violent crime. People from those places contribute quite disproportionately to the total count of victims.

The relatively extreme spatial concentration of victims, especially victims of violent crime, presents a challenge to samplers. Most surveys involve area probability samples in which successively

smaller subsets of geographical areas are selected until a requisite number of sample points are chosen for interviewing clusters of respondents. If crime is clustered on the basis of demographic, economic, and physical features of individuals and neighborhoods and is disproportionately concentrated in certain small areas, probability samples reflecting only the distribution of people in the population across space will very often miss the mark. Those who have reports to make about a large proportion of all criminal victimizations may be only sparsely represented in such samples.

Turner and Dodge (1972) point out that high crime areas may be characterized by multiple victimization as much as (or even instead of) widespread victimization. Perhaps the most difficult group to study in any population is that made up of multiple victims. They are not uncovered very often in victimization surveys, although they contribute disproportionately to the resulting estimates of victimization rates for the population as a whole. Table 1 examines the frequency of reports of multiple victimization in the National Crime

Survey. It reports the distribution of victims of two personal crimes-robbery and assault - in the data from the first 6 months of the National Crime Survey in 1977.

This analysis defines a "multiple victim" in the narrowest possible fashion as one who suffers twice or more from the same type of crime within a single 6-month reference period. Note that despite the verv large samples involved, very few persons recalled being victimized more than once by the same type of crime in a 6-month period. On the other hand, this group contributed 20 percent of all the assault victimizations (but fewer of the robbery incidents) reported in the survey. Table 1 also should make it clear that general population surveys are a very cumbersome and expensive way to study this very small group, and the sampling and measurement errors associated with their identification make it very difficult to generalize about the results.

The inability of interviewers to locate representative samples of certain subgroups in the population also is troublesome. The difficulty is that factors associated with the victimization rate frequently are related to nonresponse. U.S. Census Bu-

	·		Туре с	of crime		
Crime attribute	Interpersonal violence	Robbery	Personal theft	Burglary	Larceny	Auto theft
Percent cannot dentify offenders						· ·
Sex	3.2	3.9	46.6	94.6	95.7	92.4
Race	4.0	6.1	47.5	95.2	95.8	93.0
Age	9.5	7.9	49.0	95.2	95.9	92.9
Percent who do not know relation to	0.0	26	46.0	00 5	00.0	01.0
(N)	(3,777)	(1,023)	(512)	(5,789)	(19,601)	91.8 (1,198)

reau interviewers have been extremely successful in making contact with sample households and persons living there. Response rates for the National Crime Survev remain high (above 95 percent), even while responses to surveys conducted by academic and commercial polling units have slumped considerably. However, as Martin (1978) and others have pointed out, geographic location, lifestyle factors, and other correlates of respondent inaccessibility and problems in the management and administration of surveys often seem to go together with crime. Victimization rates are higher among transients, lower income persons, young males, city dwellers, those who recently have moved, and persons (like taxi drivers or bartenders) in occupations which often make them difficult to find.

Attempts to locate samples of victims selected from police files indicate that this group is particularly hard to cover, for completion rates in such studies have ranged from only 63 to 68 percent (Yost and Dodge, 1970; Turner, 1972a). This point is quite important, for there is a strong tendency for individuals who are victimized during one time period to report victimizations during later waves of

the National Crime Survey (Lehnen and Reiss, 1978). However, changing residence (and thus moving out of the sample) also may be a common reaction to serious and multiple victimization (Reiss, 1977c).

Furtive nature of crime

Successful criminals elude detection or commit crimes with such speed and skill that they cannot be identified or described. Thus victims often cannot reliably describe attackers or their methods. In the crime surveys almost one-half of all victims of purse snatchings say that they cannot identify the race or estimate the age of their attackers. In the case of burglary, many cannot specify (even roughly) the time the crime took place. This severely limits our ability to use surveys to identify crime patterns or to use victims' descriptions to study offenders independently of police statistics. Many aspects of crime will remain hidden from sight regardless of the methods employed to examine them.

The extent of this inaccessible information about crime is documented in table 2. In that table, data from the National

Chapter 3

Conceptual issues

Crime Survey for 1973 are employed to illustrate the frequency with which victims who were questioned responded that they did not know about selected details concerning crime incidents. The ability of victims to report in detail about their experiences varies by type of crime. In general, the amount of information which they have to share varies in direct contrast to the frequency of crimes. The highest rate of "don't know" responses is among victims of larceny and burglary, by far the most common of these crimes. Victims are most able to recall the characteristics of offenders in crimes involving direct contact of some sort between them and criminals. Hindelang (1978) has taken advantage of these data to produce profiles of offenders in robbery and assault which are independent of (but strikingly similar to) those available from the police on persons arrested. However, events with fleeting victim-offender contact, including purse snatchings, yielded surprisingly little information on the apparent age or race of the offender. For property crimes, there was little information available at all about offenders.

This problem becomes even more intractable when we consider using survey techniques to gather data on other common victimizations, including shoplifting, fraud, and crimes of stealth and deception. They have low rates of detection, and there is no simple means to account for their frequency in the absence of shopclerks spotting offenders, or consumers becoming aware that they have been cheated. It seems that there are far more offenses of this type than are known to their victims.

n .

The form taken by victimization surveys reflects decisions which were made concerning the nature of crime and the utility of various ways of knowing about it. In the National Crime Survey, victimizations are conceptualized as discrete incidents, with a beginning and an end, and sharply bounded in space and time. As a result, the survey does not measure well continuous processes which are not so clearly delineated and which resemble enduring conditions more than discrete events. Further, victim surveys usually are oriented toward incidents rather than victims. They only measure events that can be uniquely described, thus ignoring classes of crimes for which victimization is quite prevalent even though the frequency of individual incidents is unknown. Finally, victimization surveys abstract events from their social context and define criminality without reference to the assessments of those directly involved in the incident. It is not clear that this is desirable, and there is evidence that participants in the surveys impose such criteria on the data in any event. All of these conceptual positions are reflected in the manner in which the concept of victimization is operationalized in surveys.

Operationalization of victimization

Operationalization is the translation of concepts being measured into terms workable in light of the measurement technology to be employed. In the victimization surveys, this takes the form of specific questionnaire items which are designed to elicit reports of crimes from their victims. Those questions define the concepts being measured in concrete terms. BJS's survey efforts rely on the criminal code as the source for the definition of criminal events. The questionnaire items parallel guidelines developed by the FBI for its crime reporting system in identifying components of events which are used to signal the occurrence of a crime.

Respondents in the survey are not simply asked if they have been victimized. If we

ask people what bothers them, we will generate a great deal of data on the incidence of street urchins, demonstrators, noisy neighbors, and other things which lie outside the purview of the criminal law and beyond the capacity of the police to handle even in an informal manner. Rather, respondents are asked to report on their participation in specific events. The occurrence of a criminal event is indicated when they recall that they experienced or observed things resembling key elements of crimes. At the heart of the survey instrument is the "incident screen." A series of questions is asked each respondent, including:

Did anyone beat you up, attack you, or hit you with something such as a rock or bottle?

Were you knifed, shot at, or attacked with some other weapon by anyone at all?

Property crimes are probed by questions concerning the actual theft of property, or the observation of specific evidence that someone had attempted to steal something.

These items serve as "memory jogs." Each is designed to assist respondents in scanning their memories for crimes occurring within the survey's reference period. In the National Crime Survey, 17 items are employed to elicit "yes" responses from anyone who was a victim of one of those incidents. Eleven questions which are asked of all respondents are reproduced in figure 2. An additional six questions about household crimes such as burglary are administered to an adult informant in each household. At the end of the screening questionnaire, two "catch-all" questions (items 47 and 48 here) are included to stimulate the recall of any incident that has been overlooked. Respondents are asked if they had called the police about anything else that they thought was a crime, or if something had happened to them which they could have reported to the police. However, only reports of inci-

dents which could be classified as rape, robbery, assault, burglary, or theft are retained for analysis.

This procedure for operationalizing victimization implies at least three conceptual decisions concerning the nature of crime: That crimes are discrete events which are bounded in space and time, that they are knowable only as distinct individual incidents, and that they can be understood apart from their social context.

Discrete events versus continuous processes

The fundamental unit of analysis in crime surveys is the victimization: an incident involving a victim(s) and an offender(s), which has a beginning, some characteristic activity, and an end. Events that resemble this ideal can be firmly placed in space and time, enabling us to examine day-and-night, public-and-private, and seasonal cycles of criminality. This undoubtedly is a useful way to describe many criminal incidents, including street robberies, store break-ins, and simple thefts. But there are many other kinds of crime (even by the definition employed in the crime surveys) which more accurately may be thought of as continuous processes rather than discrete events. What observers count as discrete incidents may be instances of ongoing disputes, conflicts, or predations. Several crimes which recently have come to the attention of the American public fall into this category, including child abuse, spouse abuse, and robberies of children in school. A commercial crime in this category of "continuous" criminality is price fixing. Because these are more or less enduring conditions rather than discrete events, they are difficult to count in conventional fashion. Incidents in this category share all of the barriers to reliable measurement to be discussed here, in addition to their intractability with regard to conventional accounting practices.

For example, consider a family in which the father comes home drunk every night. regularly beats his wife, and threatens his

Individual screen questions		
	INDIVIDUAL SCRE	EN QUESTIONS
36. The following questions refer only to things that happened to you during the last 6 months – between1, 197 and, 197 Did you have your (pocket picked/purse snatched)?	Yes - How many timest	46. Did you find any evidence that someone ATTEMPTED to steal something that belonged to you? (other than any incidents already mentioned)
37. Did anyone take something (else) directly from you by using force, such as by a stickup, mugging or threat?	Yes - How many times?	47. Did you call the police during the last 6 months to report something that happened to you which you thought was a crime? (Do not count any calls made to the police concerning the incidents you have just told me about.)
 Did anyone TRY to rob you by using force or threatening to harm you? (other than any incidents already mentioned) 	Yes - How many times?	Wasser No - SKIP to 48 Yes - What happened?
39. Did anyon≁ beat you up, attack you or hit you with something, such as a rock or bottle? (other than any incidents already mentioned)	Yes – How many times?	Look at 47 – Was HH member 12+
40. Were you knifed, shot at, or attacked with some other weapon by anyone at all? (other than any incidents already mentioned)	Y≥s - How many. times?	CHECK attacked or threatened, or was some- ITEM C thing stolen or an attempt made to steal something that belonged to him?
41. Did anyone THREATEN to beat you up or THREATEN you with a knife, gun, or some other weapon, NOT including telephone threats? (other than any incidents already mentioned)	Yes How many times?	 48. Did anything happen to you during the last 6 months which you thought was a crime, but did NOT report to the police? (other than any incidents already mentioned) No - SKIP to Check Item E
42. Did anyone TRY to attack you in some other way? (other than any incidents already mentioned)	Yes - How many times?	Yes - What happened?
43. During the last 6 months, did anyone steal things that belonged to you from inside any car or truck, such as packages or clothing?	Yes - How many times?	CHECK Look at 48 – Was HH member 12+ 1 Yes – Hew many attacked or threatened, or was some 1 times? thing stolen or an attempt made to 1 No
44. Was anything stolen from you while you were away from homo, for instance at work, in a theater or restaurant, or while traveling?	Yes - How many times?	Do any of the screen questions contain any entries for "How many times?"
45. (Other than any incidents you've already mentioned) Was anything (else) at all stelen from you during the last 6 months?	Yes - How many times?	ITEM E P Interview next HH member. End interview if last respondent, and fill item 13 on cover page. Yes - Fill Crime Incident Reports.

Figure 2

children, who may in turn be protodelinquents in their own right. Occasionally the family may generate an official statistic, as when the wife defends herself with a kitchen knife, or when the collective noise level reaches such heights that the neighbors call the police. When a crime survey interviewer enters the scene, a different set of statistics might be generated. Neither recordkeeping system adequately captures the situation; each samples ongoing activities in selected slices of time. What is being observed is a condition rather than an event. Another example would be a business concern whose employees regularly carry off merchandise. Is each distinct exit from the store a separate crime, or is the victimization rather one which is carried out over time? Are thefts by each employee separate victimizations?

This problem is most severe in the case of "series offenses." Occasionally respondents report that a particular kind of event occurred "many times" or "every week." In an in-depth study of these reports, Dodge and Lentzner (1978) found that they could be placed in four categories. The largest group was occupationally related incidents. It included victims who were law enforcement officials, bartenders, or bus drivers. All of these incidents were assaults. The next largest group was of related-party violence cases. These incidents involved family members, neighbors, and friends. In the bulk of those reports, the same offenders seemed to be involved in repeated incidents. Next were crimes involving violence against children. Most of these incidents were school related and involved victims and offenders of roughly the same age and sex. The smallest group was the ubiquitous "other" category. Overall, series incidents were disproportionately violent crimes rather than property offenses.

In the National Crime Survey, series incidents are those that are so frequent. similar in character, or otherwise unmemorable that their victims cannot disentangle them for an interviewer. In particular, if victims cannot recall the specific months in which similar incidents occurred, they are classified as series crimes. About 100 series incidents are recorded every month in the survey, and they make up about 3 percent of all incident reports (Dodge, 1975).

It is difficult to use the data that currently are being recovered on series incidents to estimate their frequency as discrete events. By definition, respondents have difficulty recounting details about those incidents which are sufficient to place them individually in space and time. In addition, it is hard to unambiguously classify them in analytic categories. Robberies and purse snatchings are difficult to differentiate even when a great deal of information about them is available. As a

The fundamental unit of analysis in crime surveys is the victimization: an incident involving a victim(s) and an offender(s), which has a beginning, some characteristic activity, and an end....

result, series incidents currently are excluded completely when National Crime Survey data are processed to produce national estimates of victimization rates. This has a substantial impact on the magnitude of those estimates. Reiss (1978) calculates that including series incidents would increase the estimated number of crimes in the United States by 18 percent. The irony is that those victims who in some ways have the worst crime problems, partially because of the intense difficulty of their condition, are not counted as victims at all.

Events orientation

The series-incident problem is illustrative of a larger conceptual issue: whether crimes can be recorded in the victim surveys only if they are known as discrete events. There are other ways to discern crime, and there are indicators which are revealing of its presence and magnitude even in the absence of data on individual events. Often these indicators are the only way to detect its occurrence, and by following incident-based counting rules we thus exclude from consideration crimes which are known to occur and have measurable consequences.

Most crimes are undertaken covertly. They are committed by persons who want to reveal as little as possible regarding their identity and behavior. The most successful crime is one which is never discovered or which leaves such a limited or confusing residue of clues that it is difficult to discern what in fact happened. Thus, under current crime-measurement procedures (and this includes official police figures as well as those based upon surveys) a businessman who conducts an audit or an inventory and discovers a shortage of cash or merchandise can make an insurance claim but cannot report anything that becomes a crime statistic. On the other hand, if an employee is observed stealing something or if a shoplifter is apprehended, those incidents are eligible to be recorded by the police and interviewers. The difficulty in the first case is that no discrete event was observed, only indicators of its magnitude-dollar losses. Limited by the event-counting approach to crime measurement, we ignore a great deal of crime about which we could gather useful information.

An alternative approach to estimating the magnitude of crime problems of this sort would be to turn from incidence to prevalence measures of victimization. The BJS's victimization surveys focus on discrete events and yield rates of the incidence of crime. A prevalence orientation to studying victimization focuses instead on victims, and yields data on the proportion of individuals or households which have been victimized. The unit of analysis thus shifts from crimes to the targets of crime. For example, in 1968 the Small Business Administration conducted a nationwide victimization survey of business establishments. Businesses were asked to report whether or not they had suffered any financial loss from shoplifting and employee theft and if they had apprehended anyone committing those crimes (Small Business Administration, 1969). This survey found that two-thirds of all retail businesses suffered shoplifting losses and that 31 percent had apprehended an offender during the previous year (Reiss, 1969). Using as its analytic focus the "victim or not" dichotomy, the study then explored the correlates of victimization by these two types of crime. Thus, even in the absence of information on the frequency of specific incidents, a survey such as this could yield important insights into victimization.

Social meaning of incidents

The major conceptual position implied by the incident screen is that we can talk about crime apart from its social meaning, the interpretation applied to an activity by those directly affected by it. Social meanings differentiate many objectively similar events. In general, when civilians kill policemen it is a crime, but when policemen kill civilians it is not; parents who strike their children "discipline" them, while children who strike their parents are committing a crime (in commonsense terms) only if they are grown up. Teachers, on the other hand, cannot strike anyone – but a decade ago many kept paddles on their desks. The problem of social meaning is not important when we examine street robbery or anonymous assault, but physical aggression within family and friendship circles, and thefts

and robberies in which the offender personally was known to the victim, are more difficult to interpret. The "criminality" of an incident in such circumstances may depend very much on the expectations of those involved. Victimization surveys find that "it was not a police matter" is the explanation most often offered for not reporting most of those incidents to the police. The use of an incident screen which calls for reports of behavior may elicit information about activities which appear to fall within a standard crime category, but this does not always mean that it is appropriate to label them "criminal," especially if those directly concerned would not.

The meaning of an event also revolves around the issue of intent. Criminality is a concept which, in law, is not strictly behaviorally defined. Many people are injured in ways difficult to distinguish from criminal violence except through interpretations of those events supplied by participants. In a methodological study involving interviews about physical injury, Biderman (1975) guizzed over 600 people about their aches and pains. He found that 18 percent of those who were in pain or suffering from a handicap attributed their conditions to a criminal event; many more were victims of trauma inflicted by others, but those actions were interpreted as carelessness, error, or the result of inaction. Criminality is a label which may be officially conferred only after intent is established: this is what distinguishes manslaughter from murder. One may argue that from the victim's perspective the effect is the same. However, it is often necessary to establish intent in order to determine what kind of victimization an event was. For example, consider a shopkeeper who arrives at his store in the morning and discovers that his store window (still largely in place) has been smashed by a brick. What happened? It may have been an attempted burglary, a serious crime, or it may have been vandalism, which is not defined as a serious crime. The definition of the event will be determined by the interests of those involved and the procedures of those doing the measuring. The police will want to call it vandalism, for that is a Part 2 offense. Crimes in that category are rarely discussed when the FBI's statistics are announced. The motives of the shopkeeper

That we cannot agree upon what a crime is or how it may be isolated for measurement and analysis is indicative of how far victimological research has yet to proceed.

will be mixed, and his interpretation of the facts probably will be determined by how he wants to deal with his insurance company. The commercial victimization survey, on the other hand, could have recorded the incident as an attempted burglary, especially if the story gained some detail in retelling. Note that every crime measurement technique confronts the same cloudy information base. In that circumstance, social processes will determine the outcome of the investigation.

The importance of the social interpretation of events may be reflected in one of the major puzzles presented by data now produced by the National Crime Survey: the apparently inflated number of reports of victimization from assaultive violence contributed by higher status respondents. Victimologists always have assumed that the bulk of victims of assault comes from the lower reaches of the social ladder. Lower status persons are heavily overrepresented among victims of such crimes that are recorded in police files. However, one of the most universal findings of victimization surveys is that education is frequently only weakly and sometimes even positively correlated with reports of victimization by assault. Illustrative data from the National Crime Survey are presented in table 3. In this table, rates per thousand for assault are contrasted for groups reporting differing levels of educational attainment. Note that for 1976 those with college degrees recalled three times as many assaults as those with only an elementary school certificate.

This puzzling state of affairs seems ubiquitous. It characterized earlier surveys in the United States (Dodge et al., 1976), as well as surveys in Germany (Stephan, 1976), the Netherlands (Steinmetz, 1979), Norway, Sweden, Finland, and Denmark (Wolf, 1976a). If it is a methodological artifact, rather than a reflection of the "true" assault rate, it is an extremely robust one.

There are at least two competing explanations for this phenomenon. One is that it is an artifact of differential productivity in interview situations. This will be discussed in detail below. The other explanation is that respondents of different classes apply differing interpretations to

	Table 3.	
i	Rate of assaultive violence.	
	by educational attainment	

Education	Total assault rate per thousand
Elementary school 0-4 years 5-7 years 8 years	9.8 9.2 7.2
High school 1-3 years 4 years	14.3 14.8
College 1-3 years 4 years	24.8 21.5

Source: U.S. Department of Justice, 1979a, table 15. Note that this table only includes persons 25 years of age and older. The data is for reference year 1976.

certain encounters. If the "true" distribution of violence is as victimologists traditionally have assumed, those who are unaccustomed to physical abuse may find it more memorable. What to some may be a condition of daily life may seem to others a brush with criminal violence. Because crime by its nature involves imputed motives and the imposition of definitions upon events by observers, differences in what respondents remember or think interviewers are asking may greatly affect the apparent victimization rate for selected groups. To explain such findings from his study in Stuttgart, Germany, Stephan (no date) speculated that they reflected differences in "sensitiveness toward criminality" on the part of upper and lower status persons.

Evidence internal to victimization surveys does not always support this interpretation of the data. For example, if differences in assault rates by social class were strongly influenced by differential interpretations or reconstructions of encounters, we would expect the positive education-assault relationship to be stronger for more trivial events; they presumably are more amenable to differential interpretation. It is not. Likewise, we

cial standing, such as family income, to evidence similarly puzzling features. They do not: assault reports drop in frequency with increasing income (U.S. Department of Justice, 1979a:table 13). On the other hand, Stephan (1975) reasoned that "sensitivity to crime" should be greater among residents of Stuttgart than those of large American cities because crime is much less common in Germany. Therefore, Germans should remember more trivial events. He tested this hypothesis by xamining the ratio of unsuccessful to successful crimes recalled in victimization surveys conducted in identical fashion in the United States and Germany. He found that Germans were more likely than Americans to recall unsuccessful or uncompleted crimes. While not as convincing as evidence in the other direction, this explanation might account for puzzling differences among American cities. For example, assault rates were higher for residents of San Diego than New York City, even controlling for their personal characteristics (Skogan and Klecka, 1977). Individual differences in the saliency of events also might explain some perplexing racial differences in the distribution of victimization by assault. Most notably, white residents of Washington, D.C., recalled 21/2 times as many assaults as did black residents in the City Victimization Survey conducted there in 1974, and all of that difference was in the "simple assault" category (U.S. Department of Justice, 1975:table 3, p. 247).

would expect alternative measures of so-

It should be underscored that these conceptual controversies and their sometimes unsatisfactory resolution are not confined to the National Crime Survey; they reflect, rather, underlying, unresolved controversies within the field of criminology itself. That we are not agreed upon what a crime is or how it may be isolated for measurement and analysis is indicative of how far victimological research has yet to proceed.

Chapter 4

Measurement issues

Victimization surveys collect data on criminal incidents through interviews with participants. This use of self-reports of past events raises important measurement issues. Participants in a victimization survey are more akin to observers than to respondents in traditional opinion surveys. We assume that victims have been involved in events which have intersubjective meaning about which independent observers could agree. The task of interviewers is to elicit accurate reports of those occurrences.

Surveys tapping the hopes and fears or voting intentions of the citizenry strive for reliability in measurement. Because those surveys are probing internal states, researchers primarily are concerned that their readings of those states are stable and not highly dependent on a particular survey question. Victimization surveys strive for an additional goal, that of validity. Because the survey gathers data on events external to the individual, and those events presumably have a reality apart from their description to an interviewer. the standard of accuracy in victimization research is the match between the reality of an incident and its description.

This match is problematic under the best of circumstances. The problem is exacerbated by the nature of crime, conceptual disagreement surrounding the definition of criminal incidents, and a host of human processes affecting the accurate recall and description of things which occurred in the past. As a result, data on victimization may seem extraordinarily fragile, overly dependent upon subtle variations in the manner in which we gather it. In 1966 the Bureau of Social Science Research conducted the first investigation of victimization survey techniques. Its report concluded: Our survey method is heavily depend-

ent upon the ability and motivation of the respondent to remember events and report them in the interview situation. In our pretest and survey experience, we have found that the quality of the reports of victimization that are

elicited by our interviews depends to a considerable degree upon how the task of remembering and reporting is structured by the interview schedule and, presumably, by the way in which the interviewer uses it (Biderman et al., 1967:52).

Like the data on crime gathered by police, reports of victimization reflect both the distribution of events and our procedures for eliciting those reports.

Beginning with the work of the Crime Commission, there has been a great deal of research on specific techniques and strategies for improving the quality of victimization data. In addition, we can call on a substantial body of methodological research in related fields which confronts problems similar to those plaguing victimization studies. These include investigations of the quality of data gathered in surveys of unemployment, household expenditures, and health.

These studies suggest that retrospective reports of experiences with crime are clouded by four kinds of error. Each reflects fundamental human processes, and affects social measurements of all kinds. Errors in the measurement of victimization may be due to (1) ignorance of events. (2) forgetting (or not telling), (3) inaccurate or incomplete recall (or lying), and (4) differential interview productivity. In addition, there are a host of procedural problems and decisions about survey tech niques, which seriously affect the data; they will be considered in the next section. Respondents sometimes do not know of things about which we quiz them. They also might have forgotten about them, a fallibility which in practice we cannot distinguish from their deliberately not telling us about them. Respondents may also either inadvertently or malevolently tell us something that is incorrect. Finally, some people are better respondents than others: they more readily grasp the nature of the task presented them; they work harder at it; and they tire of

the demands of the survey less rapidly. All of these factors conspire to shape the volume and character of reports of victimization, sometimes independently and sometimes in conjunction with the true distribution of criminal incidents.

Methodological research techniques

Most of what we know about measurement problems in victimization surveys comes from three kinds of research. The first methodological research technique is analytic: it involves carefully examining the results of a victimization survey to infer the impact of various methodological features of the study on the data. The second technique is experimental; it involves varying specific survey methods across parallel samples and then comparing the resulting estimates of victimization rates or other aspects of the data. The third method is criterion validation: it depends on the existence of some alternative record of a crime which we can assume is accurate and we can compare to the results of an interview with the victim. Each of these techniques has made an important contribution to our understanding of the nature of error in the National Crime Survey and related efforts. The substance of those contributions will be discussed in later sections. Here we examine briefly the strengths and weaknesses of these research tools.

Analytic methods

A simple examination of data gathered in a victimization survey often is revealing of substantial methodological problems. For example, comparison of victimization rate estimates by month or quarter across the length of the recall period of a survey always reveals that reports of offenses are more frequent in months closer in time to the date of the interview. In January 1971 a segment of the sample for the U.S. Census Bureau's Quarterly Household Survey was asked about crime experiences for 1970. Eighty percent of

... the victimization survey gathers data on events external to the individual, and those events presumably have a reality apart from their description to an interviewer ...

the personal crimes that were recalled by victims took place (as they were remembered) during the last 6 months of the period (Turner, 1972b). The same "bunching" of events in more recent months has been observed in Germany (Schwind et al., 1975) and England (Sparks et al., 1977). This does not seem to reflect the true distribution of crime.

Internal analyses of victimization data also point to weaknesses in reports gathered through proxy interviews. In early surveys an "informed adult" often was quizzed to gather data about the victimization experiences of all members of a household. However, in studies for the Crime Commission both Biderman et al. (1967) and Ennis (1967) found that respondents recalled many more incidents which involved themselves than they did incidents which involved others. Finally, reports of victimization may not be related to other important measures, like race and education, in ways that seem credible. During their early stages of development we often have less confidence in our measures than in our theories, and such findings indicate that further methodological research may be called for.

As this suggests, perhaps the greatest failing of inferential methodological exegesis is that we can only guess what might be error and we can only infer what the causes of error are. We recognize the excessive clustering of crime because we generally understand the seasonal pattern of offenses, and we presume that it occurs because victims forget more easily incidents which happened further in the past (actually, that is only partially responsible for the observed clustering). Many people have been puzzled by the positive relationship between education and assault rates revealed by the National Crime Survey. The data seem wrong; on the other hand, we collected the data because we thought that the base of our knowledge of victimization was inadequate. We often do not know enough about crime to recognize to what extent an observed distribution is affected by methodological factors. For example, in an analysis of National Crime Survey data over time, Lehnen and Reiss (1978) examined the extent to which respondents who in past waves of the survey had reported crimes were likely to report them in subsequent waves. Their hypothesis was that there would be a negative effect of respondents' past experiences in completing the detailed incident report questionnaire used in the survey. When they found that victims were more likely to report crimes in subsequent interviews, they argued for a substantive finding that some people are "victimization prone."

Experiments

The second widely used technique for exploring methodological quirks in victimization data is the parallel sample experiment. Alternative forms of a questionnaire can be administered, or different procedures can be utilized across groups, to explore the consequences for the data that are collected. Strictly speaking, this is an experiment only when assignment of membership in those groups is random; this condition has been met in most of the research reviewed in this monograph.

Perhaps the best example of an experiment supporting the development of the National Crime Survey was that conducted in 1971 in Dayton and San Jose. The issue was the relative advantage of gathering victimization data through interviews with every member of a sample household as opposed to interviewing a household informant. In each city half of a household sample was completely interviewed, while in the remainder only an informant was questioned. Not unexpectedly, the former procedure produced more reports of victimization (Kalish, 1974).

Experiments using randomized assignment have been used to examine other issues as well. Cowan (1976) investigated the impact of the current practice of interviewing parents about the experiences of their children. Victimization rates were higher for youths who were interviewed personally than they were for those represented by proxy. Two supplements containing additional attitude and judgmental questions have been added to some city and national survey questionnaires. Because this was done at random, the impact of those additional questions on

estimates of the victimization rate could be analyzed. In each case, presenting additional items about crime and the seriousness of victimization before the main questionnaire seems to have stimulated the recall of more criminal incidents (Cowan et al., 1979; Balvanz, 1979).

Several investigations of the impact of survey procedures have focused on the issue of temporal telescoping. Telescoping is the tendency of respondents in retrospective surveys to "bring forward" events in time. This systematic-recall bias is a threat to victimization research because it may inflate the apparent victimization rate for the calendar period of interest, the reference period of the survey. This is controlled in the National Crime Survey by "bounding" each 6month reference period by a previous interview; data are to be collected only on events which have occurred since the previous reference period and interview. Because people are randomly assigned to panels in the survey, it is possible to compare the reports of those who are being interviewed in the same month for the first time (thus there is no bounding interview) and for the second time (a bounded interview). These two random samples are reporting on events which occurred in the same 6-month period, so it is presumed (perhaps falsely, as we shall see) that differences between their reported rates of victimization can be attributed to the effect of that initial bounding interview (Turner, 1972b; Murphy and Cowan, 1976).

The experimental nature of these investigations lends a great deal of credibility to their findings, due to the power of random assignment. There are weaknesses in this experimental approach, however. The criterion by which the "better" method or procedure is to be chosen as a result of these studies is unclear, and in the end the decision always depends on an argument based upon other information. It has usually been assumed that "more is better," that the procedure which produces the largest number of victimizations is more accurate. Tuchfarber and Klecka (1976) argued that their telephone survey procedure was better than parallel personal interviews because they uncovered more reports of victimization. However, this is not an unambiguous criterion.

There may be reasons to believe that "more" is not necessarily more accurate. Interviews have demand characteristics which shape how diligently respondents work to complete their task, and conceivably certain techniques could encourage over-reporting. It seems clear that respondents could be encouraged to telescope forward more incidents from the past, or to reinterpret trivial and perhaps marginally criminal events, or even to lie in an attempt to please an interviewer.

In addition to the absence of a criterion, these studies are limited by the large num ber of interviews which are required to test conclusively the effects of methods or procedures. The relative infrequency of crime means that either very large samples or very large method effects must be involved if a cross-sample difference is to be significant. In the proxy interview study, which was conducted as part of the San Francisco City Victimization Survey, only 570 persons 12 or 13 years of age lived in the 9,778 households in the sample (Cowan, 1976). The interviewing experiment was far from definitive, due to the large sampling error involved.

Record checks

A third procedure for identifying methodological weaknesses in reports of victimization is the record check. The U.S. Census Bureau conducted three record checks while developing the procedures employed in the National Crime Survey (Yost and Dodge, 1970; Dodge, 1970; Turner, 1972a). In each study, samples of incidents were drawn from police files, and interviewers were dispatched to quiz victims named therein. The data gathered in these interviews were compared to the official records. Two questions were examined: "Did the victim recall the incident for the interviewer?" and, "Did the victim accurately identify the month in which it occurred?" Record checks thus document the recovery power of the survey and the validity of the dating of incidents. They were used to test successive improvements in the survey's instruments and the length of the recall period that respondents could be expected to report on accurately. (These record checks are reviewed in detail in Sparks et al., 1977, and Hindelang, 1976.)

... the standard of accuracy in victimization research is the match between the reality of an incident and its description.

In addition, there has been one major record check which reversed this process. It began with victims' reports gathered in a survey and attempted to match those with records from their local police department (Schneider, 1977). While the U.S. Census Bureau's investigations were largely confined to the power and accuracy of incident reporting, Schneider's study was concerned with the interviewrecord match of descriptions of offenders, victim-offender relationships, reports of self-protective measures taken by victims, perceptions of response time, as well as other data elements.

The record-check approach to the validation of reports of victimization is potentially a powerful tool for methodological research. However, the credibility of the findings depends in large measure on three assumptions: That the record employed in the comparison contains the correct view of the event, that the findings of record-check studies can be extended to cases in which no record was generated, and that problems in fielding such studies do not influence their findings.

The first difficulty with record-check validation is the assumption that a police file is a useful criterion for judging the veracity of victims' reports of their experiences. It has been assumed that the detail employed in the U.S. Census Bureau's validation studies-the month in which the incident occurred - was correctly reflected in police records. While this seems to be reasonable, the assumption that the police and victim necessarily would classify an incident into the same analytic category, or would interpret the event in similar fashion when making their respective reports, probably is not. As we have seen above, there can be pressure on both the police and the victim to recast events. The record check in San Jose suggests that attempted rapes and assaults were particularly prone to differential classification (Turner, 1972a). Further, Schneider's (1977) record comparison found a great deal of disagreement between the record of the Portland police and victims' descriptions of key elements of events. For example, on the question

of whether or not the victim and offender were known to one another prior to the incident, the two reports agreed only 56 percent of the time. From the victim's point of view it also seemed that the Portland police were prone to classify assaults in less serious, Part 2 categories. In neither case is it clear that the police were correct in recording the crime. It should be noted as well that data from police records and police decisions regarding an incident may reflect information gathered from a variety of sources other than the victim including their own observations and reports of witnesses. This is another reason why details about events drawn from the two sources may not always be in agreement.

A second difficulty with the record-check approach to validation of incident reports is that it is limited to incidents which somehow came to the attention of a recordkeeper. The victimization surveys themselves suggest that only 50 percent of all serious crimes are reported to the police. Reported crimes are systematically different from unreported incidents, principally in terms of their seriousness (Skogan, 1976a). In general, record checks have been conducted only on crimes which are more serious and which are reported, investigated, and recorded by the police. Those crimes certainly should be the most vividly remembered by victims. Biderman (1971:4) has noted that we should expect:

... poor recall of victimization for the type of unreported incident where the victim sees nothing whatever he can do about it (except cry over spilt milk). No pattern of actions follow upon the event that reinforce its psychological impact and provide additional concrete anchors in experience for recalling it. Woltman and Cadek (1977) report that the "memory decay" curve apparent in data from the National Crime Survey is not as steep for reported as for unreported incidents or for those which are more serious. All of this suggests that record checks conducted to date probably overestimate the aggregate accuracy of the reports of victimization gathered in the surveys.

There do not seem to have been any record-check validations of victimization reports which have utilized reports other

The problems of respondents not knowing things which a measurement technique assumes they have knowledge of can have significant implications for the design of victimization surveys.

than those on file with the police. Thus we have no reading of the accuracy with which unreported crime is recalled in the National Crime Survey, despite the fact that the gathering of such data is one of the major goals of the project. Reiss (1977b) has suggested interviewing people who talked with the victims about their experiences and comparing those recollections with descriptions by victims who did not call the police, in an attempt to validate the recall of unreported crimes. Samples could also be drawn of households to which the police were dispatched, but where they did not write up an incident report. Record-check studies have the advantage of knowledge of the "true score" under investigation, which lends them great analytic power. There has not been enough critical or innovative research with regard to that criterion, however.

The final problem with record checks is the apparently universal tendency for victims to be hard to reach for interviewing. In every study of this type a substantial proportion of victims sampled from police files cannot be found or refuse to be interviewed. As a result, we are uncertain of the generalizability of the findings of these studies to the larger and apparently more transient victim population.

None of the record checks conducted by the U.S. Census Bureau has matched its usual standard for interview completions. In the city of San Jose a victimization survey of the general population enjoyed a 97 percent completion rate. As part of that project, a sample of victims was selected from police files, and their addresses were imbedded in the general sample. In that special victim group interviews were completed for only 63.5 percent. The bulk of the noninterviews (76 percent) was with people who simply could not be located; an additional 11 percent of victims moved from the city, and 13 percent refused to be interviewed or were never available (Turner, 1972a). This completion rate was the lowest of all the record checks conducted by the U.S. Census Bureau, although the figures were only slightly higher (about 68 percent) in studies conducted earlier in Baltimore and Washington, D.C. In London, Sparks et al. (1977) had even worse luck; they

could find only 43 percent of their known victim sample, and 8 percent refused to cooperate.

These low completion rates are not surprising. It is prosecutor's lore that the first response of many victims of crime is to arrange an unlisted telephone number or to move to a new address. In the National Crime Survey people who recently have moved report higher rates of victimization than those who have not, and a substantial proportion of those reporting multiple or series victimizations moves to another address prior to the next wave of interviews (Reiss, 1978; Lehnen and Reiss, 1978). In addition, many victims (and witnesses) give false addresses to the police in order to avoid further involvement in a case or retaliation by their antagonist. It is unclear how generalizable the findings based on those who remain accessible are to all crime victims.

Knowledge of incidents

The problem of respondents not knowing things which a measurement technique assumes they have knowledge of can have significant implications for the design of victimization surveys. The commercial surveys conducted for LEAA were limited in scope to burglary and robbery, deliberately avoiding the difficulties involved in gathering incidence figures for shoplifting and employee theft. Individuals also may not recognize that an incident is a crime; this has limited the utility of surveys for studying offenses such as fraud. People also seem to exclude broad ranges of their experience as lying outside of the purview of the criminal law. Respondents in the national survey conducted for the Crime Commission were encouraged to volunteer reports of victimization for crimes not explicitly covered in the interview. However, Ennis (1967) notes that few respondents mentioned ordinance violations, housing discrimination, illegal treatment by government agencies, or other such offenses

The bulk of the research concerning the problem of lack of knowledge on the part of respondents has focused on proxy interview procedures. In early surveys it was assumed that crimes were salient events that would be widely discussed, at least among members of a victim's household. Therefore it was assumed that it would be possible to conduct a victimization survey by interviewing just one adult in a household, asking him or her about the experiences of each household member. This procedure would seem to generate victimization data for a large number of individuals at low cost.

Subsequent analysis of data gathered in this fashion indicates that the method is inadequate. Biderman et al. (1967:45) found a 2-to-1 discrepancy in the frequency of reports of incidents personally involving informants and those affecting other household members. A difference of the same magnitude was found by Ennis (1967:102) in his national survey. In Biderman's Washington, D.C., study the correlation between the size of a household and the number of incidents reported there was even negative, rather than positive, in sign. In the national survey proxy problems especially biased victimization rate estimates for younger blacks, who rarely were the household member interviewed and whose experiences thus were underrepresented. The same pattern of underrecall for persons other than the respondent has been found in surveys overseas. In a survey in Stuttgart, Stephan (1976) questioned residents of 741 households. In some he interviewed all members of the family directly, while in others he interviewed only heads of households and asked them to report on victimization of other members of their family. Direct personal interviews proved to be almost 50 percent more productive of victimization reports.

Because no careful analysis of any of these data has been reported, we can only speculate about why the household informant technique seemed unsuccessful. Perhaps many crimes which went unreported using this technique were so minor they were not generally discussed. Others may have been so embarrassing that they were edited out of family conversation. This may be particularly true of crimes involving some culpability on the part of the victim, or of sex crimes. As part of its data quality control program, the U.S.

Census Bureau conducts reinterviews with small samples of respondents each month. Those who gave inconsistent responses in the two interviews are questioned about the difference. One respondent confronted with an inconsistency replied:

Mother was present when I was interviewed. I didn't dare tell (the) interviewer in front of mother. I didn't know if I really should tell about it anyway. (Graham, 1974:4)

Researchers in the past apparently overestimated the extent of communication which goes on within families, especially across age lines. As noted above, experimental personal interviews with 12- and 13-year-olds indicated that self-reporting by young respondents rather than proxy reporting by their parents yields information about more incidents, most notably assaultive violence (Cowan, 1976).

The apparent unreliability of household informants as sources of data about the experiences of others led LEAA to fund the experiment in San Jose and Dayton that was described above. In half of the households interviewed there (and the sample was 11,000 households in each city) a "chance respondent" was interviewed, and in the other half every resident 16 years of age and older was quizzed. Differences between the estimates of victimization rates produced by the two methods were substantial; the ratio was 1.7 to 1 for rape, 2.1 to 1 for strong-armed robbery, and 2.2 to 1 for attempted robbery, in favor of the self-response technique (Kalish, 1974:37). As a result of this experiment LEAA decided to adopt completeenumeration samples for city and national surveys, despite the substantially greater cost this entails.

The unreliability of household informants continues to be illustrated in National Crime Survey reports of property victimization. In addition to being quizzed about the standard litany of personal crimes, one respondent in each household is asked to supply details about such incidents as burglary which presumably affect the entire family. This supplemental household questionnaire is thus given to only one informant at each address. However, an examination of all of the

that a substantial proportion of the crimes which should have turned up in the household interview (13 percent for burglary and 30 percent for household larceny) were in fact reported by someone else in response to screen questions which were not designed to stimulate the recall of household crimes (Dodge, 1977a). An informant cannot be relied upon to supply details even about burglary or auto theft. This raises the question of how many additional household incidents would have been uncovered in the surveys if appropriate memory cues were supplied for every respondent.

Forgetting and not telling of incidents

There is a tendency for victim-respondents to fail to report information about incidents which have occurred and about which they should have been knowledgeable. We can observe examples of nonrecall in methodological studies employing each of the three research techniques described in the previous section. For instance, the way in which an interview is structured affects the frequency with which instances of criminal victimization are recalled. Experiments reveal that when respondents have to work harder at their assigned recall task, or when the task is organized so that they easily can learn how to reduce their workload, they will respond by restricting the amount of information they contribute to the survey. Second, record checks indicate that victim recall can be highly selective. Respondents seem to edit incidents which may be embarrassing or may be considered "none of the government's business," even when they previously were reported to the police. Finally, victimization rates analyzed monthly or quarterly over the length of the survey's recall period typically indicate that few incidents occur in the most distant months, although other evidence suggests that crime was just as frequent then.

incidents revealed by the survey indicates In each case the observed variations in victimization rates are artifacts of the method employed to gather the data rather than reflections of the distribution of the true rate of crime. There are three general sources of nonresponse which correspond to the examples given above: Respondent load and fatigue, purposeful suppression of valid responses, and forgetting. Because crime surveys employ verbal interviews to elicit victims' reports, it is not possible to distinguish directly between these sources of nonresponse. Physiological measurement techniques, including the use of lie detectors and devices to measure galvanic skin response, potentially could identify outright lying, but these have never been used to validate responses in victimization surveys.

Load and fatique

The effects of workload factors were first noted in the Bureau of Social Science Research's pretest of victimization survey methods. It experimented with two procedures for conducting interviews. In the first, respondents were given flash cards describing criminal incidents. If they indicated that they had been involved in such an event, a detailed incident report form was completed for it at that time. The other procedure involved asking respondents to give "yes or no" answers to a complete checklist of offense descriptions before filling out incident report forms for each positive response. The first procedure clearly linked a positive response with a lengthy respondent task, while the latter did not allow the respondent to become test wise until it was too late. Not surprisingly, the second mode elicited 21/2 times as many reports of incidents as the first (Biderman et al., 1967).

The current screening procedure used in the National Crime Survey reflects this experience. By deferring the introduction of incident forms until the completion of the incident checklist, it may encourage more complete recall. However, there may still be a tendency for respondents to suppress reports of victimization in order to speed the interview, a disposition that presumably would be greater in surveys with 12rather than 6-month reference periods.

Surveys that employ a household informant entail a considerably heavier respondent burden. Biderman (1973) speculates that once respondents have manifested their cooperativeness by recalling a victimization, there is less pressure in the interview situation to remember others, because the interviewer has been "satisfied." Personal interviews are social interactions. Interviewers ask for people's time, and they can offer little in return. Respondents may reciprocate by offering a little to the interviewer and then stopping. This may explain the surprisingly slight incidence of multiple victimization documented above. Given the average number of victimizations in the population, statistically we should find fewer nonvictims and more multiple victims than currently are uncovered in surveys (Sparks et al., 1977). Fatigue, impatience with the repetitiveness of the incident screen, and other factors may account in part for the observed distribution. This is likely to be more common among poorly motivated respondents, those who find interviews taxing or incomprehensible, and those who find few social rewards in chatting with someone from the U.S. Census Bureau. Biderman (1973) speculates that such persons may be more likely to be victimized by crime as well.

Lying and not telling

The evidence that respondents may be lying, or deliberately suppressing reports of events of which they have full knowledge, is quite inferential. It comes primarily from record checks based on reports of incidents sampled from police files. In the San Jose methodological study described above, evidence emerged that respondents who were known victims were neglecting to describe particular events. The relationship between the victim and the offender as recorded by the police seemed to play an important role in the recall of those events in subsequent interviews. As indicated in table 4, incidents in which the victim and the offender were related to one another were reported in the survey only 22.2 percent of the time. That recall rate rose sharply when the relationship between the parties was more tenuous. For events involving

Table 4. Patterns of record-check resp	onse	
Incident characteristics	Percent recalled	Number of cases
Offender a stranger	76.3	99
Offender known	56.9	78
Offender related	22.2	18
(Total cases)*	(63.7)	(206)
Assaults-total	48	81
Assaults by strangers	54	24
Rapes-total	67	45
Rapes by strangers	84	19
*Includes other subcategories. Source: Turner, 1972a:9.		

strangers the recall rate was 76 percent. Two-thirds of the personal victimizations that were not recalled involved at least an acquaintance between the parties, while three-quarters of all "stranger" crimes were recalled. Eleven of the fifteen rapes which went unmentioned involved nonstrangers.

Almost an identical pattern was uncovered in a record-check study of the validity of survey reports of assault conducted by Statistics Canada. They found that 71 percent of stranger assaults were recalled, but only 56 percent of "known party" assaults and 29 percent of relatedparty assaults were recalled (Catlin and Murray, 1979:table R). Those figures are extraordinarily similar to findings from the San Jose record check.

There are competing explanations for this phenomenon. Victims may not remember disputes which arise within kinship or friendship circles as readily as they remember events involving strangersthe data in table 4 may reflect true forgetting. Or, such disputes may not register as the kind of incidents that the interviewer

is looking for-they may not be construed as crimes. People may think that to be a "crime" violence must involve strangers. However, these alternatives seem unlikely, for these incidents all were "founded" by the San Jose police.

It may be that persons who have been victimized by someone they know frequently may not think it is any of the interviewer's business. Or, the survey may raise again the memory of a painful situation, one which victims may not wish to recall. Although these all were incidents which came to the attention of the police, the victim may not have been the party who called them; many crimes are reported to police by friends, relatives, and bystanders, and the offended party may not wish to spread the story even further. Finally, in related-party cases the question of who is to blame and who is the real victim is not always clear, and the role of the person being interviewed might not always withstand close scrutiny. It is possible that an interview with any of the participants in these affairs could have recorded what appeared to be a victimization.

Victims who are themselves culpable may also be motivated to suppress information about criminal incidents. Research

on crime indicates that "victim precipitation" is a common phenomenon in violent crime and in incidents where the victim knows the offender. In those incidents it is the eventual victim, rather than apparent offender, who first initiated the event. Other crimes may be encouraged or facilitated, if not caused, by citizen behavior. Biderman's (1967) survey in Washington, D.C., dealt in passing with this problem. There, 25 percent of all victims agreed that they were negligent or had done something foolish which contributed to their plight. Victims who feel culpable may be less likely to report their experiences later in an interview.

Forgetting

Most research on nonrecall has focused on what is assumed to be true forgetting. The problem has been described variously as "time-dependent error" and "memory decay," for it appears that the difficulty is one of remembering incidents from the more distant past. At one time it was assumed that crimes were very memorable events; it was planned to use retrospective surveys of the general population to reconstruct an historical time series for victimization rates, using interviews with a life-long reference period. Pretests quickly demonstrated the futility of that enterprise. Rather than being readily memorable, Biderman et al. (1967:31) found:

In practice, most respondents seemed to find it difficult to remember incidents of victimization other than recent cases.... People reported hours, days, and even weeks later that incidents they had not remembered at the time they were interviewed had come to mind subsequently.

In the Washington, D.C., survey, respondents were asked to recall the "worst crime that has ever happened to you." They recalled a total of 260 incidents in response, only 108 of which occurred more than 2 years previously, and only 60 of which happened 6 or more years in the past (Biderman et al., 1967:41). Biderman et al. (1967:40) noted:





Fi	a	u	re	3
	-		•	-

Respondents have to do a great deal of thinking and slow reflection before they can remember even fairly serious crimes of which they were victims some time ago-even when these older incidents are far more consequential than recent ones.

The National Crime Survey now inquires only about what has happened "in the last 6 months."

The fact that victims forget about their experiences with the passage of time also has serious implications for the accuracy of estimates of victimization rates based on surveys. This is illustrated in figure 3, which shows how different the estimates of victimization would be if they were cal-

Table 5. Record-check recall, by months of recall demanded			
Months between Interview and Incident	Percent recalled	(N)	
1-3 4-6 7-9 10-12	69 50 46 30	(101) (100) (103) (90)	
Source: Turner, 1972	2a:8.		

culated on the basis of crimes which were described as happening 1 month ago, 2 months ago, etc. If we used crimes described as occurring only 1 month ago, we would find that the national rate of victimization from personal theft was 189 per 1,000, and for all personal crimes 261 per 1,000. However, with increasing lengths of recall those estimates would have dropped sharply. Based on incidents recalled for the sixth month before the interview, the corresponding rates would drop to 119 per 1,000 for personal theft and 162 for all personal crimes.

As we shall see below, not all of this gradient can be attributed to the forgetting of past incidents. It is also shaped by forward telescoping. However, decreases in rates for personal crime and burglary of nearly 100 incidents per 1,000 over a 6month recall period clearly signal trouble. As we saw with regard to the 1971 Quarterly Household Survey, the problem is even more extreme in 12- as opposed to 6-month recall periods, and this doubtless affects the yearly victimization estimates produced in the city surveys conducted between 1972 and 1975.

Similar declines in recall with the passage of time can be observed in data from record-check studies. In record checks, samples of cases of different "ages" are drawn from police files. Interviews with victims are employed to determine if those from the more distant past are less likely to be recalled. Record checks are more definitive studies of the forgetting problem because other factors which affect the distribution of data such as that in figure 3 are not present. Table 5 summarizes the findings of the San Jose record check. It

The fact that victims forget about their experiences with the passage of time also has serious implications for the accuracy of estimates of victimization rates based on surveys.

indicates the proportion of incidents that was remembered by victims in light of the number of months of recall that they required. As table 5 indicates, recall was relatively high for cases from 1 to 3 months in the past, but it hovered around only 50 percent for those from 4 to 9 months in the past, and then dropped below onethird for those from nearly 1 year in the past. As the author of the San Jose report noted, based on this criterion "... there is very little to choose from after the first three months" (Turner, 1972a:8).

Declining rates of recall with the pressure of time also were noted in earlier U.S. Census Bureau record checks in Washington, D.C., and Baltimore, although patterns in the Washington study were less clear-cut (Dodge, 1970). In Baltimore, levels of recall were much higher than in San Jose, averaging 81 percent, but evidenced a steady decline with passing months (Yost and Dodge, 1970). On the other hand, Sparks et al. (1977) found high rates of recall (averaging 92 percent) and only a slight decline in that rate over a 10-month period.

There is considerable interest in patterns of forgetting, for they have implications for how we interpret data with such error. For example, if incidents which are reported to the police are less likely than others to be forgotten, this will increase the apparent reliability of victimization reports, because record checks have all been based on police records. On the other hand, if forgetting is unrelated to the characteristics of incidents or the attributes of victims, it is much less likely to lead us to make false inferences from the data.

Conflicting evidence

Evidence about the relation between incident characteristics and rates of forgetting is mixed. One of the first investigations of the problem was done as part of a study on a subject not related to victimization. Neter and Waksberg (1964) examined reports in a survey of household repairs and concluded that less expensive projects were likely to be forgotten more rapidly with the passage of time. The results of studies of criminal victimization

have not been consistent with this regard, however. The Washington, D.C., record check indicated that incidents involving smaller financial losses were not more likely to be forgotten (Turner, 1970). Looking at general crime categories, Ennis (1967) concluded that nonrecall in his survey was unrelated to incident seriousness. Comparing monthly official crime rates with victimization rates for eight cities, Gottfredson and Hindelang (1977) came to the same conclusion, but in a fourth study, Woltman and Cadek (1977) found that crimes involving weapons (which presumably are more serious) were less likely to be forgotten.

Similarly, studies have been inconsistent with regard to the relationship between whether or not victimizations are reported to the police and patterns of forgetting. Gottfredson and Hindelang (1977) found that monthly survey rates for reported and unreported crimes deviated from official figures for eight cities in the same fashion, indicating that there was no relationship between notifying the police and remembering the incident later in an interview. On the other hand, Woltman and Cadek (1977) noted that reported crimes evidence less of a time-dependent gradient than unreported crimes; this is similar to Schwind et al.'s (1975) data for Göttingen, Germany, Note, however, that these analyses necessarily involved the inspection of patterns of reports of victimization and not record checks.

Only record-check data can speak definitively about memory decay, for reports of victimization (and other events) in retrospective surveys are strongly influenced by telescoping as well. While telescoping can be both forward and backward in time, the net effect of these two forces often is strongly in the forward direction (Schneider, 1977). There is a significant tendency for respondents to "move" events around in time, falsely describing them as being more recent than they actually were. In a set of victimization data, forward telescoping (which shifts events into later months) and forgetting (which primarily affects earlier months) combine

to produce memory error gradients like those in figure 3. In the end, forgetting is a much more significant threat to researchers than is telescoping, for in the latter case events are recalled with inaccuracy rather than being overlooked completely. Telescoping within a reference period will not affect data on victimization for groups, nor will it affect estimates of city or national rates. Forgetting, on the other hand, will certainly affect estimates of victimization rates, and it may lead us astray in fundamental ways if it is related to the attributes of crimes or victims.

The inextricable relationship between forgetting and telescoping in victimization data recommends further record-check research. With only two exceptionsdollar value of losses and victim-offender relationships-none of the U.S. Census Bureau's reverse record checks has examined correlates of nonrecall other than the passage of time. In England, Sparks et al. (1977) concluded that there were few attributes that differentiated between those who did and did not recall known victimizations. In general, nonrecall was not related to age, race, residential mobility, sex, social class, or the attributes of incidents. On the other hand, because most incidents in the London study were recalled, there was little variance to be investigated.

More recall research is necessary to inform decisions about the optimal length of the reference period to be employed in victim surveys. The National Crime Survey currently employs a 6-month recall period, although we have seen that both record checks and visual inspection of the resulting data suggest that substantial time-dependent memory error occurs even over that length of time. In Scandinavia, where relatively little methodological research has been done on these matters, victimization surveys typically employ reference periods of 2 years or even longer (Wolf, 1976b).

In terms of time-dependent forgetting, the best recall period is the shortest one. People doubtlessly can give the most accurate information about events which occurred "yesterday." Other factors also affect decisions about the length of a survey's reference period, however. Brief reference periods will generate relatively little data on victimization for most individuals, especially about serious crimes. As a result, there is a tradeoff between length of reference periods and the size of the sample required to make stable estimates of victimization rates. Also, errors in measurement due to factors such as telescoping will have a greater proportional effect on data gathered for shorter reference periods (Reiss, 1977b).

Inaccurate or incomplete recall of incidents

The failure of respondents to share information about events which apparently did involve them is not the only type of error encountered in data gathered in interviews with crime victims. Information which is volunteered may be incorrect or at least different from that gathered on the same incident from other, presumably more reliable, sources. Victims make mistakes: they may inaccurately recall the amount lost in a crime or the date of the incident. In their London study, Sparks et al. (1977) compared the month in which victims placed criterion incidents with police information on the same offenses. They found that only 55 percent recalled the month of the offense with accuracy. Victims may also deliberately misconstrue their role in a crime, the value of a stolen object, or the identity of an offender. This may be common in crimes that involve close victim-offender relationships, victim complicity, or victim precipitation. The police often suspect the motives of complainants, and so might survey interviewers.

Record-check comparisons of archival and interview data indicate that at least two types of recall error present serious methodological problems: temporal telescoping and misreporting. There can be a great deal of disagreement, some of which appears to be time dependent, about the characteristics of offenses and offenders between these two data sources.

Telescoping

The issue of temporal telescoping has received a great deal of attention, because it has profound implications for survey design and cost. In an early study, Gray (1955) conducted a record check of reports of sick leave by British civil servants. He found that few forgot completely that they had taken leave, but that there was a substantial tendency for them to err in recalling when they took it. Neter and Waksberg (1964) investigated telescoping problems in self-reports of household repairs. They found that recall error was predominantly in the forward direction, moving events closer to the date of the interview. They also discovered that major expenditures-which presumably would be more memorable- were more likely than minor expenses to be telescoped forward. Because minor expenditures also were more likely to be forgotten, the error structure of the U.S. Census Bureau's self-report data on household repairs was very complex.

Although telescoping is found in many retrospective surveys, it is not altogether clear why it takes place. Part of it may be due to the demand characteristics of a victimization survey. In this case, the "demand" to produce an incident occurs because most respondents have not been victims of most of the crimes covered in the interview. The long incident screen produces a succession of "no" responses, and respondents may feel the interviewers are "disappointed" by their lack of productivity. In this situation, the temptation to give the interviewer some false but apparently satisfying information may be overpowering when a familiar but slightly out-of-bounds incident comes to mind (Biderman, 1970). There is also some evidence that frequent and recurring events are telescoped more often, for there is a greater likelihood that the respondent will become confused about his or her dates (Sudman and Bradburn, 1974). The "interview demand" hypothesis does not explain, however, continued forward telescoping even within the reference period for a survey, a phenomenon

noted by Neter and Waksberg (1964) and in all of the victimization record checks.

For purposes of making accurate estimates of victimization rates for a calendar period, any disposition by respondents to draw into the reference period events which took place before (or after) it is more threatening than errors in time placement within the period. The more threatening phenomenon is known as "external telescoping." Various survey techniques have been developed to deal with this problem. One solution to this has been to "bound" surveys conducted for estimation purposes by an earlier interview. The bounding interview, which takes place at the beginning of the reference period, gathers reports of prior incidents and serves as a benchmark for the ensuing timespan. Interviews conducted at the conclusion of the reference period presumably are then protected from forward telescoping. In addition, incident reports gathered in the initial interview can be used to screen later interviews to eliminate duplications. Another aid to recall is to shorten the length of the reference period and to locate its terminal point as close in time as possible to the date of the interview. This increases recall accuracy (the demand for details about temporally distant events is eliminated) and limits the scope for backward telescoping. The trade-off, of course, is cost. Finally, external telescoping can be reduced by "bounding" the beginning of the reference period with a salient date. During interview pretests for the National Crime Survey it became apparent that people had difficulty locating events in time because of the absence of salient reference points. They appeared to remember incidents which occurred in January more frequently than many other months because they "came just after the first of the year" (Yost and Dodge, 1970). Interviews which refer to reference periods with natural boundaries marking their beginning and end seem to be more satisfactory.

19

Controlling telescoping within a reference period is much more difficult than controlling external telescoping, and there is no effort to do so in the National Crime Survey. There are, however, procedures which serve to enhance recall accuracy in retrospective surveys which are applicable in this case. Even in the complete absence of a bounding interview it may be possible to increase the accuracy with which respondents pinpoint the timing of events by assisting them in the construction of their own mental benchmarks. In a victimization survey in England Sparks et al. (1977) opened their interviews with a series of questions about respondents' recent activities. Londoners were asked about "things which you did in the past year," including vacations, major illnesses, births, marriages, and job changes. Exact dates were solicited for each of these events, which are usually highly salient in the minds of the participants. When victimization information was gathered later in the interview, the incidents were put within the context of these events. It would be ascertained, for example, that an assault took place between a holiday to Brighton and the marriage of a daughter. This appeared to increase the accuracy with which the dates of criminal incidents were elicited, for the known victim subsample in this survey (selected in the same fashion as those in the record checks) exhibited less telescoping than American victim samples.

The effects of external telescoping on victimization rate estimates can be considerable. In the 1970 Washington, D.C., record check, for example, some individuals were selected for interviewing because police files indicated that they had been victimized 7 months before. They were asked only about their experiences during the "past 6 months." About 15 percent of those out-of-bounds incidents were pushed forward into the reference period. Over 20 percent of a sample of 13month-old cases were incorrectly placed within a 12-month reference period by another group of victims (Dodge, 1970). In July 1971, the issue was investigated experimentally. A victimization instrument was administered to 18,000 participants in the Quarterly Household Survey,

	Rate estimates per thousand				
Type of victimization	Bounded panel	Unbounded panel	Percent* difference		
All personal crimes	64.3	84.0	31		
All violent crimes	16.4	22.7	39		
Personal thefts	47.9	61.2	28		
All household crimes	103.2	137.6	33		
Burglary	42.8	57.1	33		
Vehicle theft	9.0	11.8	32		

on unrounded figures. Source: Woltman, Bushery, and Carstensen, 1975, Table 3. Data are for January to June 1973 recall period.

12,000 of whom had been interviewed about crime in January of the same year. The survey asked about their experiences "in the past 6 months." In every crime category, the 6,000 respondents whose interviews were unbounded reported more incidents than those who had been questioned before. The ratio of unbounded to bounded reports ranged from 1.2 to 1 for burglary to 1.9 to 1 for robbery. This was roughly the same magnitude of error due to telescoping in Neter and Waksberg's (1964) comparison of bounded and unbounded reports of household repairs: their unbounded interviews yielded 40 percent more reports of expenditures.

The design of the National Crime Survey enables us to make comparisons between bounded and unbounded interviews on a continuous basis. Table 6 reports on differences between estimates of the national victimization rate for several crimes based on interviews conducted for repeat, bounded samples and new, unbounded samples for the reference period January to June, 1973. In every case, the new, unbounded households entering the survey reported more instances of victimization than those which were already part of the sample used for estimation purposes; in the aggregate, the difference in rates was about 33 percent, a very substantial discrepancy attributable to this single methodological difference.

It should be noted that the data presented in table 6 actually understate the impact of bounding on telescoping. Many interviews in the "bounded" column in table 6 were in fact unbounded. In theory, individuals who are interviewed for the first time are questioned to establish an initial reference period and to gather data for use later to eliminate duplicate reports of events. In practice things are more complex. The U.S. Census Bureau conceives of the survey sample as a sample of addresses. Thus it treats an address as bounded when at least one person there has been interviewed. As a result, many respondents are treated as bounded when they have not been interviewed before. This group includes others at the address who were not interviewed when it first entered the sample, new persons in that household, younger household members who "age" into the sample, panel participants who were missed on a prior round of interviews, and - most important-entire households who move into a sample

address at some point after the bounding interview (Jacob, 1975). In an exhaustive analysis of several years of data from the National Crime Survey, Reiss (1978) concluded that in most periods between 17 and 19 percent of those interviewed were falsely considered bounded—there had been no interview 6 months before. The bulk of these interviews was with members of "replacement households" which had recently established themselves at the address.

The impact of the inclusion of unbounded interviews in the data used for estimation purposes is considerable. Reiss (1978) reports that fully one-third of all victimizations uncovered in the survey are reported in unbounded interviews. This is partially due to external telescoping and partially to the fact that people who move generally report higher rates of prior victimization than those who do not. Although the bounding procedure in principle controls for the former, the failure to implement it fully plays a large role in determining the apparent level of victimization in the United States.

In this regard it is also important to note that the city victimization samples interviewed by the U.S. Census Bureau were unbounded. The interviews conducted in 26 cities between 1972 and 1975 employed 12-month unbounded reference periods. In eight of the cities, the reference period also did not refer to a calendar year (January through December), which probably further detracts from the quality of the data. We do not know enough about the consequences of this to predict its impact on other measures. If more serious incidents were telescoped into the reference period (Reiss, 1978) while less serious ones were more rapidly forgotten (Neter and Waksberg, 1964), the relative mix of crimes as well as rates of victimization were affected. On the other hand, external telescoping should have proportionally less of an impact on reports gathered for a 12-month period than it does on the unbounded components of the National Crime Survey with its 6-month reference period. Respondent fatigue and forgetting should be greater over the longer span, however.

In addition to reconceptualizing bounding procedures, research on the telescoping process should focus on internal telescoping effects and on the correlates of telescoping itself. We know little about why events are telescoped or about their differential misplacement in time. In her record check in Portland, Schneider (1977) examined the kinds of events which were most severely moved about in time. Her survey employed a 12-month reference period. On the average, matched incidents were pulled forward within that period by 2.2 months. Forty-nine percent of all incidents were placed in the wrong month by their victims. She found a weak tendency for more trivial incidents to be telescoped forward more often, and for events which occurred more distantly in the past to be pulled forward more frequently. Also, crimes in which the victim reported resisting the offender often were misplaced in time. However, the tendency to move events forward in time was not related to the age, race, sex, or educational level of respondents.

Telescoping within a reference period presents analytic difficulties, for it impedes our understanding of the timing and sequencing of criminal incidents. Even within the 6-month reference period currently employed in the National Crime Survey, survey incidents apparently are being pulled forward in dramatic fashion. Twenty-eight percent of all incidents now are being placed in the first month of any recall period, four times as many as in any last month (Reiss, 1978). This destroys the utility of the data for examining issues such as the sequencing of multiple victimizations or the impact of recent experiences with crime on a victim's willingness to resist another attack or to report ensuing incidents to the police. Without accurate data on the temporal placement of incidents we cannot link them in causal fashion to other events, such as quitting a job, moving to another address, installing a crime-prevention device, or getting a divorce. To document the causes and consequences of crime at the microlevel we need accurate data on the relative time placement of many events in people's lives, including victimization.

Other sources of measurement error

Research on inaccurate recall has focused almost exclusively on the time placement of individual incidents. However, there is reason to suspect that victims are likely to recall inaccurately other aspects of events. Research in experimental psychology indicates that errors in recall increase as a function of the logarithm of time (Sudman and Bradburn, 1974), Record checks which match significant characteristics of incidents across police files and victim interviews would shed a great deal of light on the general reliability of the data collected in the surveys. The only record check of the characteristics of incidents that has been made by the U.S. Census Bureau focused on differences in estimates of dollar losses between victims and the Washington, D.C., police. That comparison revealed that citizens had substantially higher estimates of the value of their stolen and damaged property than did the police. Threefourths of the loss estimates gathered in interviews were higher than those recorded by the police, often by 50 to 100 percent. On the other hand, there was no indication that these differences were time dependent or that the dollar amount of a loss affected the accuracy of its recall (Turner, 1970).

In the Portland record check, Schneider (1977) compared police and interview data on a variety of incident attributes. She found that survey estimates of loss and seriousness consistently were higher than police figures. Victims were much more likely than police reports to mention that weapons were involved in a case. Police reports and victims also disagreed much of the time on the race of the offender and, as noted above, on victim-offender relationships. Victims also reported substantially longer response times by the police than official records indicated. On the other hand, there was a good match for such factors as the age and sex of suspects and the number of offenders involved in the incidents. Interestingly, these mismatches were not consistently related to the passage of time. Some of the incidents were from 12

Other factors contributing to the omission of offenses during an interview may not be time dependent. Record checks strongly suggest that incidents involving close relationships between the parties are withheld from interviewers.

months in the past, yet none of the error in those comparisons (scored as measures of the difference between victim and police reports) was time dependent. Also, the passage of time was not related to the tendency of the victims to give "don't know" responses to questions about their experiences. Only knowledge of the date of the incident seemed to fade with time. It would seem that the criterion of accuracy employed in the survey pretest record checks was the most stringent of choices.

Differential productivity of respondents

Research on general survey methodology indicates that respondents also differ in their willingness or ability to adopt a productive role during an interview (Sudman and Bradburn, 1974). In general, more highly educated respondents are more cooperative, more at ease in interview situations, and more able to recall the details of events. Those factors may affect the accuracy with which victimizations are recalled during interviews.

As we noted above, it is assumed that most forms of criminal victimization are more frequent among lower status persons. However, surveys conducted for the Crime Commission found victimization to be positively related to measures of social class. The strongest social class correlate of victimization was education. College-educated respondents recalled victimizations at a higher rate than did others. This surprising pattern may be due to differing definitions of victimization and attendant variations in the probability that events will be recalled in an interview. On the other hand, researchers suspect that significant negative associations between social position and victimization are masked in the survey findings by greater interview productivity among more highly educated and test-wise respondents. Higher levels of education (but not income) measure entry into a "test and measurement culture" in which surveys, questionnaires, and opinion polls are recognized features of life. In addition, more educated respondents may

enjoy greater verbal fluency of the kind necessary for conducting a bureaucratic encounter, and they may generally be more inclined to trust the stated intentions of inquiring government agents. Interviews with such respondents should be less perfunctory, involve greater task comprehension, and elicit more effort in completing the task than those with less comprehending or less able respondents.

There is also little evidence of the dimensions of the problem or of the credibility of this explanation for observed variations in victimization reports. In England, Sparks et al. (1977) found that among upper class respondents victimizations which were recalled were more likely to be trivial ones, or attempted rather than successful crimes. Similar findings have been reported for Germany (Stephan, 1976) and the United States (Biderman et al., 1967). In the National Crime Survey those proportions fluctuate considerably among those with lower levels of educational attainment, but are by far the highest for those with college training. In data collected during the first 6 months of 1977, 63 percent of all college-educated assault victims fell in the "attempted assault without a weapon" category; for everyone else that figure was 49 percent (author's computation).

The only other evidence that differences in the ability of victims to complete the interview task are affected by education was reported by Reiss (1978). He found that less educated respondents were more likely to recall incidents that fell into the "series" category, which is composed of crimes for which discrete details could not be remembered. On the other hand, Schneider (1977) found in her record check that education was not related to any tendency for victims to give "don't know" responses or to systematic differences between police reports and interview data on incidents. Based on this evidence, it seems that productivity effects must be only of the "recalled or not" variety and thus at work only in the screen section of the survey instrument. It remains unclear why nonrecall error ever should be distinct from errors in the detailed incident descriptions gathered in the incident report section of the instrument.

Summary

Conceptual and measurement decisions have had a substantial impact on the volume and nature of crime in America revealed by victimization surveys. Between 1973 and 1978 trendlines for the major categories of offenses measured by the National Crime Survey were flat, revealing little increase in crime over that 6year period (U.S. Department of Justice, 1979b) This stability highlights the importance of conceptual and measurement problems, for they have had more impact on the apparent level of crime than all the events of the 1970's.

The use of proxy respondents serves to depress the apparent rate of victimization. When people are asked to recall events for others as well as for themselves, their own experiences predominate. In addition, using a household informant places a taxing burden on a respondent, and sheer fatigue may become an important factor shaping his or her productivity. Proxy effects also may account for the sharp increase in victimization rates currently recorded between the ages of 13 and 14, the point when the National Crime Survey shifts from proxy to selfresponses for youths. In addition, proxy respondents are questioned to gather data about people whom U.S. Census Bureau interviewers cannot arrange to interview individually. Under many circumstances these hard-to-reach household members may be persons whose lifestyle would lead us to expect more frequent self-reports of victimization (Hindelang et al., 1978).

There is also evidence of considerable memory bias in the surveys, part of which is reflected in nonresponse and part in misresponse. It is useful to distinguish between the two, for there is some evidence that they reflect different recall problems. However, in a set of victimization data the two are inextricably linked. Without additional information it is impossible to disentangle time-dependent forgetting from internal forward telescoping, the two processes which conspire to produce apparently decreasing rates of victimization over the length of a reference period.



A summary of the impact of forgetting and telescoping is presented in figure 4. It plots a hypothetical "actual" victimization rate and one observed in a typical victimization survey. If a survey is unbounded, we expect to see an upward "tail" at the beginning of the reference period reflecting incidents which occurred before it began and are telescoped into it. The tendency of victims to forget events more distant in the past and to move them forward erroneously in time when they do recall them then predominates. Only in the recent past, within 3 months of the date of the interview (Turner, 1970), does the ability of victims

- to place properly criminal offenses in time approach 90 percent. Finally, the National Crime Survey and other surveys inevitably collect a number of interviews some time after the close of the reference period. This allows for the shifting of some events into the period which in fact occurred after its conclusion.
- Other factors contributing to the omission of offenses during an interview may not be time dependent. Record checks strongly suggest that incidents involving close relationships between the parties are withheld from interviewers. However,

the definition of those events as noncriminal may be an important problem as well, especially when they were not reported to the police. Experiments with various interviewing techniques indicate that respondent load and fatigue factors are quite consequential in retrospective studies and should be greater in surveys with longer reference periods or greater coverage of categories of victimization.

Evidence on inaccurate recall of the details of incidents is based on one recordcheck study. In it, temporal telescoping was again found to be a major problem, but error in the reporting of other aspects of offenses varied considerably with mag-

Chapter 5

Procedural issues

nitude. Except for the date of the event, none of that misrecall was time dependent. Police and victims disagreed on the seriousness of crimes, the value of losses, the identity of offenders, and other aspects of what happened for other reasons, most of which are not clear. However, the independence of that disagreement from the personal attributes of victims performing the recall task should be heartening to the analyst.

Our very sparse data on differential respondent productivity are quite puzzling. On the one hand, education seems to be related to self-reports of victimization in unexpected fashion. On the other hand, education is not related to inaccurate recall of the detail of events or the tendency to give "don't know" responses to questions about that topic. This is consistent with most research on the quality of reporting in surveys, which indicates that measurement error is not strongly associated with demographic characteristics of respondents. Less educated respondents are more likely to generate reports of series incidents, which is congruent with the productivity hypothesis. But otherwise the presumed productivity effect seems confined to "remembering or not." It does not seem to lead to errors or failures of recall in events that are dredged up from the past.

The results of victimization surveys also are affected by procedural factors. These are difficulties inherent in fielding national retrospective surveys, the manner in which surveys are carried out on a dayto-day basis, and variations in survey procedures which arise in different places. In this chapter we consider three of the most important of these factors. First, we will examine some of the consequences of the panel design of the National Crime Survey, including panel bias and attrition. Second, we will explore the implications of variations in modes of interviewing, comparing telephone and in-person methods. Finally, we will review evidence concerning interviewer bias in victimization surveys. There are considerable gaps in our knowledge of some of these topics. but enough data are at hand to suggest that the manner in which the surveys are being conducted has considerable consequences for the picture of crime in America which emerges from the end product.

Panel bias and attrition

The National Crime Survey is a panel study. As a result, the findings are affected by a variety of panel effects that lead to systematic biases in victimization data. Panel biases are artifacts in data attributable to the fact that respondents are visited again and again in a survey. These biases may arise because respondents tire of the task and either suppress reports of things the interviewer wants to know or fail to exert the effort necessary to succeed at it. Panel bias also arises when participation in a survey affects the attitudes or behavior of respondents. In one-time, cross-sectional surveys this presents no difficulty if those effects come later, but in panel efforts it will affect subsequent readings of the sample.

Panel bias

Panel effects seem ubiquitous. In a review of the topic, Bailar (1975) reports that participation in prior interviews has

24

depressed subsequent reports of household expenditures, repairs, and alterations. In addition, when the reports of new respondents to a survey are compared with those who are being reinterviewed at the same time, it was found that the recall of recent illness and of unemployment is lower among experienced survey participants. All of these effects seem to be attributable to panel fatigue. In Bailar's (1975) study incoming panel members reported rates of unemployment which were 20 percent higher than those reported by experienced respondents. (However, it may be that these differences are attributable to telescoping, for a new panel member is also "unbounded.")

In the National Crime Survey, respondents are scheduled to be interviewed seven times, at 6-month intervals. This is comparable to the Current Population Survey. in which respondents are to be quizzed eight times. Woltman and Bushery (1977a) took advantage of this survey design to compare victimization reports for groups with differing degrees of panel experience who were being interviewed in the same month. They found generally declining rates of reported victimization as the number of times that respondents had been interviewed increased. The largest drop was between the second and third interview, when reports of various personal crimes declined between 4 and 10 percent.

Lehnen and Reiss (1978) tackled the problem using special tabulations for individuals which merged their responses over a number of waves of the panel. They argued that panel participants could have contradictory effects: participants might become fatigued by continued interviewing and restrict their output as a result, or their ability to recall events could be enhanced by this unexpected continuing interest in their experiences on the part of the Government. Their data did not

speak unambiguously to these alternatives, however. They found that the number of times respondents were interviewed was negatively correlated with reports of victimization. But they included the results of bounding interviews in this analysis and external telescoping doubtlessly greatly affected the outcome. They also found that reporting past victimizationsand thus learning of the connection between a "yes" response on the incident screen section of the interview and a difficult respondent task — was positively related to reporting victimizations in later interviews. We have seen that this could have several interpretations, but it does argue against strong test-wise task avoidance effects in the panel.

There has been much less research on the issue of how participation in a survey affects the subsequent behavior of respondents. Surveys which employ a panel design are particularly threatened by this form of bias. The issue is not that of accurate measurement but that panel participants are no longer representative of the population from which they are drawn if such effects are present. The Center for Political Studies at the University of Michigan conducts a bivearly election study using such a panel, and it has found to its distress that repeated interviewing has increased the proportion of panel members who are registered to vote (Traugott and Katosh, 1979). It confirmed that effect by a record check of voter registration lists. The only evidence on this issue available for the National Crime Survey relies entirely on the reports of participants. Those data indicate a substantial panel effect on one aspect of victim behavior. Experienced respondents are much more likely than first-time respondents to recall that they reported crimes to the police (Murphy and Cowan, 1976).

An important unanswered question is whether or not panel participation affects rates of victimization among panel members. The linkage between participation and victimization would be indirect,

26

through the possible impact of panel membership on the adoption of self-protective measures. There is considerable correlational evidence that certain riskreduction tactics—like staying home can reduce individual and household victimization rates (cf. Skogan and Maxfield, 1980). If repeated questioning about crime, or interviews inquiring about the seriousness of various offenses, has had an impact on how people comport themselves during the following 6 months, the surveys may be underestimating the true level of victimization in the Nation.

Panel attrition

Biases in panel data also may reflect selective attrition in panel membership. Any panel study which extends over a 31/2-year period inevitably will be threatened by the loss of participants. (Twenty percent of the American population moves each year.) Even panel studies which vigorously attempt to follow relocating families are hard pressed to preserve their original sample. The National Crime Survey panel currently is organized as a study of residents of sample addresses. No attempt is made to retain contact with departing households or individuals. As a result, a considerable proportion of the people initially interviewed at those addresses does not participate throughout the life of their panel. Those who replace them at that address are treated as "replacement households" and represent their dwelling units (in an initially unbounded state) in ensuing rounds of interviewing.

A problem arises because this residential mobility and the subsequent "replacement" procedure is far from random. In particular, it appears that moving to another address is a common response to victimization. Reiss (1978) first noted that a substantial proportion of victims interviewed as part of the National Crime Survey did not reappear in data collected at their address 6 months later. Less than two-thirds of those reporting a victimization remained in place, a figure substantially lower than that for nonvictims. The more victimizations one reported, the higher the chance that he or she would move or refuse to be interviewed later.

After reporting one victimization, 24 percent of all respondents "disappeared" 6 months later, most of them along with their entire household. For those reporting three or more victimizations, 35 percent of their households subsequently vanished from the panel sample. The data indicate that victimization rates decline with continued participation in the panel (Lehnen and Reiss, 1978). In light of selective attrition it is likely that this reflects the consequences of crime for movement out of the sample rather than crime's true distribution in the population.

Telephone versus personal interviews

While the National Crime Survey and the victimization surveys conducted in 26 cities are personal-interview studies, in each case a substantial proportion of interviews was conducted via the telephone. In the National Crime Survey, contact with a sample household is initially established by the personal visit of an interviewer. During this visit the interviewer lists each household member; at that time he or she also interviews all available respondents. However, the interviewer exercises discretion about whether to complete the remaining interrogations by other personal visits or by telephone, and is to choose the easiest and most costeffective method (U.S. Census Bureau, 1979).

We do not have a reliable reading of the consequences of this procedure. Some comparisons of the results of interviews conducted personally and by telephone indicate that there are few differences between them. Comparisons of parallel surveys that have been conducted using the two methods sometimes indicate a similar equivalency, but sometimes favor one of the interviewing modes. No truly definitive experiment has been conducted detailing the consequences of mode of interview for data on victimization. Research on related topics also provides no clear lesson for the victimization surveys. There are reasons to suspect that telephone interviews may be less productive

than those conducted in person, and there are counterarguments which support the use of the telephone. The best research on the topic can be read to support both conclusions.

It is widely argued that surveys of the general population achieve higher completion rates when the interviews are conducted in person. Because the U.S. Census Bureau pursues a mixed-mode data collection strategy to pursue individual noncompletions, it is impossible to talk about the relative effectiveness of each in the crime surveys. In its Health Interview experiments, the Survey Research Center of the University of Michigan found a 10-percent difference between both completion rates and refusal rates which favored the in-person strategy (Cannell et al., 1979). On the other hand, the Center achieved virtually identical results in another study of the two techniques (Groves, 1979). Those who favor personal interviews also argue that "the data are better" when collected in that way because of the greater rapport that can develop between interviewer and respondent. Also, in intimate settings, interviewers can supply more verbal and nonverbal cues to shape respondent behavior, and both parties may be more satisfied with the emotional rewards of face-to-face contact. Comparisons by mode of interview indicate that respondents and interviewers are less satisfied with telephone interviews (Groves, 1979; Cannell et al., 1979). Respondents tend to supply less detail in response to openended questions given over the telephone (Groves, 1979). They also are more likely to evidence response-set bias, using the same verbal category in answer to a string of questions more frequently when interviews are conducted by telephone (Groves, 1979). It also seems that respondents in telephone surveys are less certain of the sponsorship of those studies or of the use to which the data will be put. Rodgers (1976) and Groves (1979) both found they were less likely than those being interviewed in person to supply sensitive personal information such as family income. In the Groves study, telephone respondents also were more likely to report that they felt "uneasy" discussing selected topics.

... enough data are at hand to suggest that the manner in which the surveys are being conducted has considerable consequences for the picture of crime in America which emerges from the end product.

Vigorous arguments can be made in support of telephone surveys as well. Some have argued that telephone interviews may be more productive because they are anonymous. On the telephone it may be possible to be more candid and matter-offact about embarrassing issues, and it may be easier for respondents to admit less desirable behavior. In a record check of the two modes of interviewing, Rodgers (1976) found that telephone reports of whether respondents were registered to vote were more accurate. On the other hand, Groves (1979) found no difference between the two approaches in terms of the social desirability of responses to various measures. Telephone interviews may often be more discrete, for other members of a household usually cannot hear the questions asked. Because the work is conducted at a central site, telephone interviewing can be better supervised than can field visits. As a result, interviews may be more standardized. Rodgers (1976) found that interviewer styles were more uniform and interviewer effects on data were less pronounced over the telephone, and that across two waves of interviews responses by members of her telephone panel were more consistent. Interviewers are undeniably safer in telephone surveys, often a significant concern. Groves also found that completion rates for urban areas were higher for telephone than in-person surveys.

Evidence on the relative validity of data gathered in each way is important, for the mixed-mode data collection strategy employed in the crime surveys is distributed in a decidedly nonrandom fashion. During the first few years of the National Crime Survey about 25 percent of all interviews were conducted over the telephone (National Research Council, 1976), and persons who are interviewed by telephone are more likely than others to be young, male, and black. All other evidence suggests that these are among the most likely groups to be victimized. If telephone interviews are not as productive of reports of victimization as those conducted in person, the resulting rate estimates will be severely affected.

Research results

There have been four major studies of the problem in the context of measuring

victimization. While they came to somewhat different conclusions, the best of them supports the use of the telephone. In the first study, the results of interviews with households in which maximum effort was made to employ only personal visits were compared with those in which telephones were used whenever feasible. No major differences between victimization reports from the two samples were apparent (Turner, 1977; Woltman and Bushery, 1977b). In another analysis, Tuchfarber and Klecka (1976) contrasted the results of parallel victimization survevs. They uncovered more victimizations over the telephone than the U.S. Census Bureau measured in person. However, Reiss (1978) analyzed several years of national data organized in panel fashion and - controlling for a host of other factors-found that telephones were 50 percent less productive than personal interviews.

The results of this research can only be labeled inconclusive. Rodgers (1976), Groves (1979), Klecka and Tuchfarber (1974), and others agree that there were few differences in relationships between variables gathered in differing fashion. The problem seems to be one of threats to the precision of estimates of the number of victimizations and other objects of interest. In 1966 Biderman et al. (1967) attempted to use the telephone to gather victimization data, but quickly abandoned the technique as inadequate. Since most recent and systematic evidence on the issue is ambiguous, additional research should be conducted if only because of the very large contingent of telephone respondents in the current crime panel. In the U.S. Census Bureau project, many persons in each "experimental" condition actually were interviewed by another mode. In the Klecka and Tuchfarber study two different organizations had conducted the parallel surveys and different sampling frames were employed for each, leaving room for a host of differences between the surveys in addition to the way in which interviews were conducted. Reiss' data are correlational and suffer from a lack of random assignment of mode of interview. Clearly an experiment is called for in which samples of individuals would be randomly

assigned to groups and interviewed in different ways, in conjunction with a second check to provide an independent reading of what responses "should be."

This design was employed by Statistics Canada in a major methodological study of the validity of survey reports of victimization (Catlin and Murray, 1979). They sampled 1,525 crime victims from the files of the Edmonton, Alberta, police department. Victims were randomly assigned into two groups, one to be interviewed only by telephone and the other only in-person. An additional random sample of adults was drawn from the city directory and mingled with the two victim samples in order to disguise the true purpose of the survey from the interviewers. This insured that respondents would often have no victimizations to report. Parallel surveys were then conducted to assess the completion rates, costs, and accuracy of recall associated with each survev method. Statistics Canada found that telephone interviews were significantly less expensive to conduct-including an allocation of administrative expenses and other overhead costs, telephone interviews cost 70 percent less than in-person interviews. Telephone interviews also were as successful as in-person efforts at reaching respondents; the completion rates of the two parallel surveys were quite similar. Finally, there was virtually no difference between the two surveys in the proportion of known victimizations which were successfully registered in the interviews. The in-person interviews recovered 64 percent of the criterion incidents, and the telephone interviews 63 percent. As a result of this experiment, Statistics Canada employed telephone interviews in its large-scale 1979 study of criminal victimization in Vancouver, British Columbia.

Interviewer effects

In addition to panel artifacts and biases related to mode of interviewing, differences among survey interviewers in the way they carry out their task also shape the resulting data. Interviewer effects are but one of several sources of "correlated response variance" (Bailar, 1976). These effects manifest themselves as variance on indicators which is shared among re-

Assessment

spondents who were quizzed by the same interviewer. The effects of interviewers can be quite substantial, especially when survey personnel have had comparatively little training and are only minimally supervised. Interviewer effects reached epidemic proportions in the 1960 Census of Population, a technical rationale for accepting a cost-cutting move to self-enumeration by use of a mail survey in the 1970 Census. In the city victimization surveys conducted by the U.S. Census Bureau, interviewer effects were comparable in magnitude to sampling error. For example, for Baltimore it is necessary to multiply estimates of sampling variance by 1.60 to calculate confidence intervals which take account of both sampling and interviewer variance. The rate for all victimizations there was 110 per 1,000, with a sampling-error range (with 95 percent confidence) around that estimate from 40 to 180 per 1,000. Taking into account the effects of correlated response variance extended that range to from 20 to 200 per 1,000. Those differences-become even more extreme when we examine particular crimes (Bailar et al., 1977).

The sources of interviewer effects are numerous. Interviewers differ in how they interpret individual survey items and in their understanding of the purpose of the enterprise. Some probe for detailed comments more vigorously than do others, and some interviewers readily accept "don't know" and other nonresponses. Interviewers also differ in how they interpret and record responses to questions and how they explain individual items to respondents who do not understand them. Often they do not link the verbal and nonverbal cues they give respondents to any productive effort on the respondent's part, thus rewarding unacceptable task behavior (Cannell et al., 1979).

Examination of the types of incidents for which interviewer effects are most substantial suggests that they involve the particularly sensitive topics probed by the victimization surveys. The most systematic analysis of those effects indicates that they are greatest for crimes in the "assaultive violence without theft" category—that is, for rapes, intrafamilial disputes, and public brawling (Bailey et al., 1978). Dodge and Lentzner (1978) noted that reports of series incidents often are first recorded when a new interviewer takes responsibility for a household. Presumably some interviewers are better than others at eliciting reports of "conditions" rather than events, while others more quickly tire of attempting to untangle vague or complex incidents.

Precise estimates of the magnitude of interviewer artifacts in the data are based on "interpenetrated sample" research. In each of the eight cities studied by the U.S. Census Bureau in 1975, interviewers were assigned batches of 80 sample households. A portion of these were randomly assigned from a pool of households distributed between pairs of interviewers. Then, comparisons were made in the data collected from these households, examining contrasts among interviewers. The analytic question was, How much of the observed variance in reports of victimization could be attributed to interviewers rather than to sampling variance and the true distribution of crime (Bailey et al., 1978)?

There was considerable disparity in reports of victimization among interviewers, between interviewers assigned to the same supervisor, and across cities. Interviewer effects were most extreme in Newark, where it is necessary to multiply estimates of sampling variance by 2.4 to take this additional source of error into account. Interviewer effects were most extreme for assaults and petty theft. Hearteningly, they were not linked to the attributes of the respondents themselves (Bailey et al., 1978). As a result, such effects will have fewer consequences for tabulations of relationships in the data. Also, the impact of interviewer variance on a set of data goes down as the number of interviewers in a study increases and the average number of respondents each one deals with decreases. Thus interviewer effects are much less significant in the National Crime Survey (Bailar et al., 1977). Conversely, telephone surveys typically employ only a few centralized interviewers, and the impact of differences among them will thus be more substantial.

Current state of the art

The National Crime Survey shoulders a difficult task, that of measuring the extent of a complex social process. Most sample surveys confine themselves to more manageable topics: they elicit information about the attributes of respondents, or they inquire about simple behaviors. The victim surveys examine confrontations between persons. Those interactions are complex and may take numerous forms. They are subject to various interpretations by their participants, and those interpretations are often commingled with the willingness of the actors involved to remember or report them to an interviewer. The experience of the Bureau of Labor Statistics in devising a measure of a seemingly straightforward concept such as unemployment suggests the difficulty of probing such phenomena. It took decades of research effort to arrive at the current (and still controversial) procedure for examining labor force participation, and in the process researchers were forced to drop all efforts to tie the concept to interpretations by respondents of their own status. People's assessments of their own willingness to work have been replaced by inquiries about specific jobseeking behaviors, an approach not unlike that adopted in the crime surveys (De Neufville, 1975).

The varieties of human experience are endless. The National Crime Survey has chosen one reasonable approach to imposing some order upon that rich empirical domain. Methodological criticisms of the effort impose a strict standard upon the data, that of criterion validity. Validity is a question of the relationship between two distinct measures of the same variable; if different nonsurvey measurement procedures identify (in this case) the same events or victims, we are more confident that the data are not artifactual, generated by the measurement process itself. However, few of the measures commonly employed in survey research

have any known validity. Survey data which are good by standards of the profession usually display, at most, internal consistency, and are related in expected ways to benchmark attributes or attitudes of the respondents. (This is known as construct validity.) The validity of selfreports even of simple behaviors is often quite low: claims about having voted often are inflated by 10 or 20 percent (Traugott and Katosh, 1979; Clausen, 1968; Weiss, 1968). In one study 47 percent of the sample misrecalled whether or not they gave money to a Community Chest drive (Cahalan, 1968). Any socially approved behavior will be claimed by more persons than actually practice it. Biderman and Reiss (1967) summarized a study which reported that 30 percent of a sample of persons known to have visited a doctor within 2 weeks prior to the interview failed to report the event, and that 7 percent of a sample of recently hospitalized persons exhibited similar lapses in memory. Forward and backward telescoping affect the reporting of vacation and sick leave and self-reports of household expenditures.

The purpose of this review is not to expound on the difficulties of conducting survey research, but to suggest that a bounded survey using an interview schedule with an overall recovery power of 75 percent or better is well within the normal range of the instruments of social science.

This assessment reflects a more general position about data—they always contain errors. Data are indicators of the relative distribution of some conceptual variable across a population. The numbers themselves only partially reflect that distribution (their true score component). They also are partially artifacts of the measurement method (their method component), and they are clouded by random noise from a variety of sources. In dealing with data the issues always are: Is the error component of the data truly random with respect to the relationships I am investigating? Am I being led astray by misinterpretations encouraged by method effects? The more we know about a set of data, the more confident we can be when we answer these questions.

Victimization research has come a long way in this regard. As a result of the methodological research described in this volume we know a great deal about errors in victimization data. The measurement of predatory personal crime, stranger violence, and serious property crimes appears to be satisfactory. The recovery power of the survey instrument is relatively high for most crimes in these categories, and interrespondent differences in interview productivity and interviewer effects are a less serious problem in these areas as well. Thus, despite all of their difficulties, the data generated by the crime victimization surveys have enormous potential for clarifying many issues in criminology.

The largest problem area remains the data on assault. While the complexity of victimization survey data demands that we interpret all the data with care, the methodological shortcomings of the enterprise all seem most to affect reports of interpersonal violence. We have seen in record checks that many assaults are not picked up in personal interviews, even when they have already been reported to the police. In the Baltimore method test, only 36 percent of all assaults were recovered in the interviews, and less than one-half of those were placed in the correct month by their victims (Yost and Dodge, 1970). In San Jose, 48 percent of the victims of assault recalled the event, but that percentage dropped to 22 percent among those who were victimized by acquaintances or members of their own family (Turner, 1972a). It was apparent in the pretests that the interview process was not eliciting thorough accounts of

... despite all of their difficulties, the data generated by the crime victimization surveys have enormous potential for clarifying many issues in criminology.

interpersonal violence and that the problem was acute in the case of nonstranger assault

We also have seen the unexpected relation between education and reports of assault victimization, a relationship which leads us to suspect that more educated respondents are most likely to remember such offenses, to define them as crimes, or to cooperate in their reconstruction in interviews. Series victimizations, which ordinarily are excluded when data from the National Crime Survey are analyzed, almost all involve assaultive violence. This leads to the severe undercounting of assaults and greatly reduces the apparent frequency of multiple violent victimization. Because series offenses are more likely to be reported by less educated respondents, they further cloud the relation between education and victimization violence. Interviewer effects also hit hardest at events in this category, overwhelmingly in the direction of undercounting them (Bailar et al., 1977; Graham, 1974). Finally, panel attrition also probably disproportionately reflects assault victimization. Especially when the incidents involve neighbors or related parties, violent assault should propel victims to seek refuge in other domiciles.

These assault-linked shortfalls in the data seem to be reflected in a number of puzzling aspects of victimization research. They undoubtedly account for the high proportion of assaults attributed to strangers in the national panel and in the city studies. Across the 26 cities where surveys have been conducted, an average of 70 percent of all interpersonal violence was attributed to strangers; in the 1973 national data the figure was 60 percent (author's computation).

Interpersonal violence

These figures do not correspond with what is known about the dynamics of interpersonal violence. The evidence from other sources is that a much higher proportion of assaults, and even rapes, takes place within friendship and family circles. Numerous studies of police homicide files suggest that strangers account for only about 25 percent of all urban murders. Homicide and assault are similar in origin and process, differing primarily in their

outcome-which is often a function of such factors as the caliber of gun employed or the availability of a doctor (Zimring, 1972). Curtis' (1974) survey of official records in 17 major cities found that only 21 percent of all assaults in 1966 were attributed by investigating officers to strangers. These proportions are similar to those revealed in numerous crimespecific studies of police file data. They render the survey data even more suspect because we also believe that violence between friends and relatives is less likely than stranger violence to be reported to the police. That police files contain approximately 31/2 times more acquaintance violence than revealed in interviews does not add to our confidence in the validity of the survey findings.

Another bit of evidence relating to the reliability of the data on violence is more qualitative, but equally persuasive: the data do not square with what the police do. On a Friday night in any big city a large proportion of squad car dispatches are made in response to complaints about domestic disturbances. Frequently these do not lead to an arrest, and rarely do the participants end up in court, but the police often are called on to defuse potentially dangerous confrontations within families and to restore peace in the immediate environment.

There are several other method artifacts which shape data on aspects of assault victimization. Method artifacts probably account for the fact that in the National Crime Survey data victim-offender relationships do not appear to affect the rate at which victimizations are reported to the police. This is a surprising-and puzzling-finding. Given the low proportion of violence within close interpersonal networks which surfaces in the survey data, that which does is probably of such a character that it is also readily reported to the police. Second, the relative dearth of nonstranger offenses in the data undoubtedly increases the proportion of assaults and rapes which was reported to have involved victims and offenders of different races. To the extent to which people are likely to know or live near persons of the same race, the underestimation of violence among acquaintances

will skew the data in favor of interracial crime. Given the potentially unsettling social consequences of high levels of interracial crime in America, artificially high reports of the rate are unwelcome.

Error in the measurement of interpersonal violence which is related to the differential productivity of respondents may account for the observation that blacks recall far fewer reports of minor assault than do whites. The most trivial form of violence in the crime survey is "attempted assault without a weapon," which includes incidents which resulted in no injury and in which no weapon was brandished. That a crime even occurred is inferential, and most of these events may better be described as threatening encounters. There is no particular reason to expect blacks to experience fewer of these episodes than whites; in fact, given what we know about class- and culturelinked youthful exuberance it is more plausible to expect the opposite. Yet in the 1973 data, fully 47 percent of all the assaultive violence recalled by whites fell into this category, while only 31 percent of the assaults reported by blacks were trivial. Almost 60 percent of all black assault victimizations were categorized as aggravated assault (involving serious injury or the use of a weapon) while only 37 percent of all attacks on whites were aggravated. This is highly unlikely. Much more plausible is the hypothesis that blacks reported fewer of their less threatening encounters, while whites dredged up everything in memory. Further investigations of the correlates of respondent productivity are required before we can make any confident statements based upon much of the data on assault.

The hypothesis that the experiences of blacks are substantially undercounted in the victim data accounts for one of the most anomalous patterns appearing in the city data when the data are used in aggregate form. At the city level, using the 26 surveyed communities as the units of analysis, we find that one of the best predictors of rates of violence is "percent white." The higher the proportion of the population which is white, the higher the rate of interpersonal violence. Because data on race are highly correlated at the city level with other social indicators, including those measuring the extent of poverty, educational failure, and the

quality of life, we also find that high rates of interpersonal violence are positively associated with good housing, low population density, high income, and high levels of formal education. This is quite unlikely. It seems rather that white communities were represented by samples of white respondents, and that they produced more exhaustive reports of events and a more thorough recounting of essentially trivial events.

Future developments

This review of the methodological development of the National Crime Survey suggests a number of specific research tasks. Some of them will require broadgauged experimental trials of alternative data collection techniques. Others will require record checks of the reports of victims and witnesses of crime. These studies are necessary if the findings of victimization surveys are to achieve the same general acceptance as other social indicators, such as the Current Population Survey (unemployment, household composition) and the Health Interview Survey (medical expenses and demands on health resources). These surveys have a long history of research and development. The Health Interview Survey, for instance, began in 1957 and has achieved a maturity which the National Crime Survey can as yet only aspire to.

Some research should focus on response validity. Record checks can be used to validate improved techniques for gathering reports of dollar losses for determining the attributes of offenders and calculating indicators of crime seriousness. Some efforts also must be devoted to the validity of the data concerning whether or not incidents are reported to the police. These data frequently are used to "correct" official crime reports in local jurisdictions (cf. Schneider, 1976; Skogan, 1976b), but we have no knowledge of the accuracy of reports of police notification.

Significant amounts of response error seem to have social sources. We need to examine more closely the underreporting of related-party crimes. If this underreporting can be traced to the differential definition of events, then appropriate

changes on how the survey's task is described to respondents and in the memory jogs supplied in the incident screen may lessen the problem. If the difficulty turns out to be that victims recall such incidents but refuse to share them with interviewers, then we may experiment with random response techniques and other strategies for granting respondents greater anonymity and confidence in the confidentiality of the survey. Finally, the issue of differential interview productivity is an important one. The positive relationship between education and victimization clouds the analysis of the race and social class correlates of experience with crime. Improved interview techniques may assist less educated respondents to perform properly the difficult recall task the survey now demands.

One of the major components of the future research agenda for the National Crime Survey concerns the optimal length of the recall period. The shorter that period the larger the size of the sample which is required for the survey. The survey now is very large, yet there is evidence that there is significant nonrecall during months 4 to 6 of the reference period. One response to the problem could be to shorten the recall period, and thus further decrease the dimensions of respondent burden. The cost implications of this are substantial. This lends even greater importance to the development of alternative techniques for facilitating more thorough recall in retrospective surveys. In particular, using respondentdefined anchoring events scattered throughout the recall period seems to enhance the ability of respondents to remember crimes, as well as to more accurately place them in time (Sparks et al., 1977). This technique also would enable us to gather better information on the timing and sequencing of victimizations and other significant events during the recall period. Because some of those events may be read as consequences of crime, this would increase the general analytic utility of the data.

The coverage of the survey (in its broadest sense) also is a lively issue. Currently the National Crime Survey confines its

coverage only to selected predatory crimes and some forms of assaultive violence. Many other forms of harm or potential risk do not fall within the scope of the survey, including obscene and threatening telephone calls and vandalism. Each of these, of course, presents added definitional problems, but surveys in other nations have indicated that both are widespread problems. In Australia obscene and threatening calls present substantial multiple victimization problems (Australian Bureau of Statistics, 1975), while in Vancouver, British Columbia, vandalism was 37 percent more frequent than burglary (Corrado et al., 1979). Expanding the coverage of the survey would not only improve the utility of the data, but it doubtless would enhance the overall recovery power of the questionnaire. As Biderman (1970a) has noted. "people experience incidents, not offenses," and "neither nature nor the minds of respondents packages events neatly in accordance with Uniform Crime Report offense classes." The more respondents are encouraged to ruminate about their experiences, the more offenses of all kinds they are likely to recall, based on linkages that may not be logical from a bureaucratic perspective. We see evidence of this in the lack of correspondence between screen-questionnaire items and the kinds of crime they currently stimulate victims to remember. It also is reflected in studies of the impact of adding new crime-related questions to the basic survey instrument; more stimuli concerning crime served to enhance recall almost regardless of their specific content.

Series incidents

The current treatment of series incidents is indefensible. By definition series offenses occurred at least three times, and the description of the latest of them must be clear enough to classify it as falling within the purview of the National Crime Survey. Not to count them at all when generating estimates of victimization rates is difficult to justify. In addition, there are many important questions about patterns of multiple victimization which cannot be resolved without better data on events of this type. For example, whether

References

or not offenders are the same across series incidents would be extraordinarily revealing of the character of multiple victimization (cf. Reiss, 1977a, 1978).

By their nature we will never be able to distinguish clearly between many of the incidents these series reports represent. It appears that vigorous interviewer training and supervision can reduce the frequency with which they are recorded. Series incidents declined in number during the first years of the National Crime Survey, probably due to increasing interviewer experience with the survey (Reiss, 1978).

The inadequate representation of series offenses in the data used for estimation purposes partially explains the apparent paucity of multiple victimization in the population. Better handling of the data on such incidents would greatly increase our understanding of victim proneness. However, multiple victimizations do not necessarily take place within neat 6month packages, and a thorough portrait of the sequential connection of events requires linking reports from individuals in the National Crime Survey across successive reference periods. This is beyond the current capacity of the U.S. Census Bureau's data system for the National Crime Survey.

Procedures for measuring multiple victimization also mitigate against finding it. Currently we discern that a type of crime has affected an individual more than once via responses to the question "how many times?" which is posed following a positive response in the incident screen. This is a very weak operationalization of the concept of crime-specific victim proneness, one which probably encourages overreporting of series events. Further, the bulk of the research done on multiple victimization has employed data from the city victimization surveys. The 1-year reference period for those surveys seems more productive for studying multiple events. However, with longer reference periods the problem of respondent fatigue also should affect the apparent frequency of multiple victimization, biasing estimates in a downward direction.

The issue of panel attrition is intimately linked to the fundamental conceptualization of the National Crime Survey. If the survey were organized around individuals, the movement of households and individual household members would trigger followup efforts. The current address orientation of the survey precludes following mobile respondents for a reasonable period of time, greatly limiting the utility of the panel data which presumably it is designed to collect. It also leads to the current definition of a "bounded interview," which has little to do with the social-psychological process of telescoping which it was designed to protect against. Selective panel attrition appears to affect estimates of the level of victimization made from the survey, and limits the utility of the data for studying one extreme reaction to crime, namely, moving elsewhere.

Finally, there needs to be a great deal of research on the implications of telephone interviews for the National Crime Survey, both for the data as they are currently collected and for the future organization of the program. As the cost of conducting the survey mounts, there inevitably will be pressure to convert to a telephone survey. Many local victimization surveys are now conducted by telephone (see Abt Associates, 1977; Skogan, 1978; Statistics Canada, 1979). We know little about the implications of the use of the telephone for undercoverage, nonresponse, and response error. We should be prepared to speak to the costs and benefits of the use of telephone interviews at the national level.

These and many other issues are now being considered by the U.S. Census Bureau and the Bureau of Justice Statistics. A large-scale project to redesign much of the National Crime Survey has been underway since 1979, directed by Albert Biderman of the Bureau of Social Science Research, Inc., in conjunction with researchers and users of victimization data from throughout the nation. The research to be conducted by this group should illuminate many aspects of the data which already have been collected in the national and city surveys, as well as serve as a model for conducting future victimization surveys.

Abt Associates (1977)

3

Victimization in Joliet and Peoria, Illinois. Cambridge, Mass.: Abt Associates.

Australian Bureau of Statistics (1975)

General social survey – crime victims. Canberra: (A.B.S.#) Catalogue No. 4105.0. June 22, 1979 (May).

Bailar, Barbara A. (1975)

"The effects of rotation group bias on estimates from panel surveys." *Journal of the American Statistical Association* 70 (March): 23-30.

Bailar, Barbara A. (1976)

"Some sources of error and their effect on census statistics." *Demography* 13 (May): 273-286.

Bailar, Barbara A., Leroy Bailey, and Joyce Stevens (1977)

"Measures of interviewer bias and variance." Journal of Marketing Research 14 (August): 337-343.

Bailey, Leroy, Thomas F. Moore, and Barbara A. Bailar (1978)

"An interviewer variance study for the eight impact cities of the National Crime Survey cities sample." *Journal of the American Statistical Association* 73 (March):23-30.

Balvanz, Bill (1979)

"The effects of the national survey of crime severity on the victimization rates determined from the National Crime Survey." Washington, D.C.: Demographic Surveys Division, U.S. Census Bureau, memorandum, 15 October.

Biderman, Albert D. (1967)

"Surveys of population samples for estimating crime incidence." Annals of the American Academy of Political and Social Science 374 (November):16-33.

Biderman, Albert D., and Albert J. Reiss, Jr. (1967)

"On exploring the 'dark figure' of crime." Annals of the American Academy of Political and Social Science 374 (November):1-15.

Biderman, Albert D., Louise A. Johnson, Jennie McIntyre, and Adrianne W. Weir (1967) Report on a pilot study in the District of Columbia on victimization and attitudes toward law enforcement. U.S. President's Commission on Law Enforcement and Administration of Justice, Field Survey I. Washington, D.C.:

U.S. Government Printing Office. Biderman, Albert D. (1970a) "Memos concerning National Crime Survey developments." Washington, D.C.: Bureau of Social Science Research, Inc., memoranda,

9 April and 10 June.

Biderman, Albert D. (1970b)

"Time distortions of victimization and mnemonic effects." Washington, D.C.: Bureau of Social Science Research, Inc., memorandum.

Biderman, Albert D. (1971)

"Memo concerning National Crime Survey developments." Washington, D.C.: Bureau of Social Science Research, Inc., memorandum.

Biderman, Albert D. (1973)

"Memo concerning National Crime Survey developments." Washington, D.C.: Bureau of Social Science Research, Inc., memorandum, March.

Biderman, Albert D. (1975)

A social indicator of interpersonal harm. Washington, D.C.: Bureau of Social Science Research, Inc.

Booth, Alan, David R. Johnson, and Harvey M. Choldin (1977)

"Correlates of city crime rates: Victimization survey versus official statistics." *Social Problems* 25:187-197.

Buckhout, Robert (1974) "Eyewitness testimony." *Scientific Ameri*can 231 (December):23-31.

Cahalan, Don (1968) "Correlates of respondent accuracy in the Denver validity survey." *Public Opinion Quarterly* 32 (Winter):607-621.

Cannell, Charles F., Lois Oksenberg, and Jean M. Converse (1979)

Experiments in interviewing techniques: Field experiments in health reporting, 1971-1977. Ann Arbor: Institute for Social Research, University of Michigan.

Catlin, Gary, and Susan Murray (1979) Report on Canadian victimization survey methodological pretests. Ottawa: Statistics Canada.

Cirel, Paul, Patricia Evans, Daniel McGillis, and Debra Whitcomb (1977)

Community crime prevention program-Seattle, Washington, Washington, D.C.: U.S. Government Printing Office.

Clausen, Aage R. (1968) "Response validity: Vote report." Public Opinion Quarterly 32 (Winter):588-606.

Corrado, Raymond R., William Glackman, and Ronald Roesch (1979)

Interim report one: Extent and distribution of victimization. Burnaby, B.C.: Department of Criminology, Simon Frasier University.

Cowan, Charles D. (1976)

"Twelve and thirteen-year-old interviewing experiment." Washington, D.C.: Statistical Research Division, U.S. Census Bureau, memorandum, 8 April. Cowan, Charles D., Linda R. Murphy, and Judy Wiener (1979)

"Effects of supplemental questions on victimization estimates from the National Crime Survey." Paper presented at the Annual Meeting of the American Statistical Association, Washington, D.C.

Curtis, Lynn A. (1974)

Criminal violence: National patterns and behavior. Lexington, Mass.: Lexington Books.

De Neufville, Janet (1975) Social indicators and public policy. New York, N.Y.: Elsevier.

Dodge, Richard W. (1970)

"Victim recall pretest—Washington, D.C." Washington, D.C.: U.S. Census Bureau, memorandum, 10 June.

Dodge, Richard W., and Anthony G. Turner (1971)

"Methodological foundations for establishing a national survey of victimization." Paper presented at the Annual Meeting of the American Statistical Association, August.

Dodge, Richard W. (1975)

"Series victimizations: What is to be done?" Washington, D.C.: Crime Statistics Analysis Staff, U.S. Census Bureau, memorandum, 31 October.

Dodge, Richard W., Harold Lentzner, and Frederick Shenk (1976)

"Crime in the United States: A report on the National Crime Survey." In Wesley G. Skogan (ed.), *Sample surveys of the victims of crime*. Cambridge, Mass.: Ballinger, pp. 1-26.

Dodge, Richard W. (1977a)

"Analysis of screen questions on the National Crime Survey." Washington, D.C.: Crime Statistics Analysis Staff, U.S. Census Bureau, memorandum, 22 December.

Dodge, Richard W. (1977b)

"A preliminary inquiry into series victimizations." Washington, D.C.: Crime Statistics Analysis Staff, U.S. Census Bureau, memorandum, July.

Dodge, Richard W., and Harold R. Lentzner (1978)

"Patterns of personal series incidents in the National Crime Survey." Paper presented at the Annual Meeting of the American Statistical Association, San Diego, Calif., 14-17 August.

Ennis, Philip (1967)

Criminal victimization in the United States: A report of a national survey. U.S. President's Commission on Law Enforcement and Administration of Justice, Field Survey II. Washington, D.C.: U.S. Government Printing Office.

Garofalo, James, and Michael J. Hindelang (1977)

An introduction to the National Crime Survey. Washington, D.C.: National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration. U.S. Department of Justice.

Gottfredson, Michael R., and Michael J. Hindelang (1977)

"A consideration of memory decay and telescoping biases in victimization surveys." Journal of Criminal Justice 5:202-216.

Graham, Dorcas (1974)

"Reasons for differences in the number of crime incidents reported on the original and reinterview survey by type of crime, Novem-ber 1972 to June 1975." Washington, D.C.: Statistical Methods Division, U.S. Census Bureau, memorandum, 30 October.

Gray, Percy G. (1955)

"The memory factor in social surveys." Journal of the American Statistical Association 50 (June):344-363.

Groves, Robert M. (1977)

"A comparison of national telephone and personal interview surveys: Some response and nonresponse differences." Paper presented at the Annual Meeting of the American Association for Public Opinion Research.

Groves, Robert M. (1979)

"Actors and questions in telephone and personal interview surveys." Public Opinion Quarterly 43 (Summer):190-205.

Hindelang, Michael J. (1976) Criminal victimization in eight American

cities. Cambridge, Mass.: Ballinger.

Hindelang, Michael J. (1978)

"Race and involvement in crimes." American Sociological Review 43 (February):93-109. Hindelang, Michael J., Michael R. Gottfred-

son, and James Garofalo (1978) Victims of personal crime. Cambridge,

Mass.: Ballinger.

Jacob, Herbert (1975)

"Crimes, victims and statistics: Some words of caution." Evanston, Ill .: Department of Political Science, Northwestern University, unpublished manuscript.

Kalish, Carol B. (1974)

Crimes and victims: A report on the Dayton-San Jose pilot survey of victimization. Washington, D.C.: National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration, U.S. Department of Justice.

Klecka, William R., and Alfred J. Tuchfarber (1974)

"The efficiency, biases, and problems of random digit dialing." Paper presented at the Annual Meeting of the American Association for Public Opinion Research, June.

Klecka, William R., and Alfred J. Tuchfarber (1978)

"Random digit dialing: A comparison to personal surveys." Public Opinion Quarterly 42 (Spring):105-114.

Lehnen, Robert G., and Albert J. Reiss, Jr. (1978)

"Some response effects in the National Crime Survey." Victimology 3 (No. 1-2):110-124.

Martin, Elizabeth (1978) "A twist on the Heisenberg principle-or

how crime affects its measurement." Chapel Hill, N.C.: Institute for Research in Social Science, University of North Carolina, unpublished paper.

Murphy, Linda R., and Charles D. Cowan (1976)

"Effects of bounding on telescoping in the National Crime Survey." Paper presented at the Annual Meeting of the American Statistical Association, Boston, Mass., 23-26 August. National Research Council (1976)

Surveying crime: Report of the Panel for the Evaluation of Crime Surveys. Washington, D.C.: National Academy of Sciences.

Neter, John, and Joseph Waksberg (1964) "A study of response errors in expenditures data from household interviews." Journal of the American Statistical Association 59 (March):17-55.

Reiss, Albert J., Jr. (1969)

Field survey: Appendix A to Crime Against Small Business. Washington, D.C.: U.S. Government Printing Office.

Reiss, Albert J., Jr. (1977a) Final report for analytical studies of victimization using National Crime Survey Panel

data. New Haven, Conn.: Institution for Policy Studies, Yale University. Reiss, Albert J., Jr. (1977b)

A note on optimal reference period for recall. New Haven, Conn.: Institution for Policy Studies, Yale University, memorandum.

Reiss, Albert J., Jr. (1977c) The residential mobility of victims and repeat victims in the National Crime Survey. New Haven, Conn.: Institution for Policy Stud-

ies, Yale University. Reiss, Albert J., Jr. (1978)

Final report for analytical studies of victimization by crime using National Crime Survey Panel data. New Haven, Conn.: Institution for Policy Studies, Yale University.

Research Triangle Institute (1977)

Analysis of the utility and benefits of the National Crime Survey. Research Triangle Park, N.C.: Research Triangle Institute.

Rodgers, Theresa F. (1976) "Interviews by telephone and in-person:

Quality of responses and field performance. Public Opinion Quarterly 40 (Spring):51-65.

Schneider, Anne L. (1976) "Victimization surveys and criminal justice system evaluation." In Wesley G. Skogan (ed.). Sample surveys of the victims of crime. Cambridge, Mass.: Ballinger, pp. 135-150.

Schneider, Anne L. (1977)

The Portland forward records check of crime victims: Final report. Eugene, Oreg .: Institute for Policy Analysis.

Schwind, Hans-Dieter, Wilfried Ahlborn, Hans Eger, Ulrich Jany, Volker Pudel, and Rudiger Weiss (1975)

Dunkelfeldforschung in Göttingen 1973/ 1974. Wiesbaden: BKA-Forschungsreine.

Shichor, David, David L. Decker, and Robert M. O'Brien (1979)

"Population density and criminal victimization." Criminology 17 (August):184-193.

Skogan, Wesley G. (1974) "The validity of official crime statistics: An empirical investigation." Social Science Quarterly 55 (June):25-38.

Skogan, Wesley G. (1975)

"Measurement problems in official and survey crime rates." Journal of Criminal Justice 3 (Spring):17-32,

Skogan, Wesley G. (1976a)

"Citizen reporting of crime: Some National Panel data." Criminology 13 (February): 535-549.

Skogan, Wesley G. (1976b) "Crime and crime rates." In Wesley G.

Skogan (ed.), Sample surveys of the victims of crime. Cambridge, Mass.: Ballinger, pp. 105-120.

Skogan, Wesley G., and William R. Klecka (1977)

The fear of crime. Washington, D.C.: American Political Science Association.

Skogan, Wesley G. (1978)

The Northwestern University Center for Urban Affairs random digit dialing telephone survey, Evanston, Ill .: Center for Urban Affairs, Northwestern University.

Skogan, Wesley G. (1979) "Crime in contemporary America." In Hugh Graham and Ted Robert Gurr (eds.), Violence in America. Beverly Hills, Calif .: Sage Publications, chapter 14 (second edition).

Skogan, Wesley G. and Michael G. Maxfield (1980)

Coping with crime: Victimization, fear and reactions to crime in three American cities. Evanston, Ill .: Center for Urban Affairs, Northwestern University.

Small Business Administration (1969)

Crime against small business. A Report of the Small Business Administration Transmitted to the Select Committee on Small Business, United States Senate. Washington, D.C.: U.S. Government Printing Office.

Sparks, Richard, Hazel G. Genn, and David J. Dodd (1977)

Surveying victims. New York, N.Y .: John Wiley,

Statistics Canada (1979)

Greater Vancouver crime survey documentation package. Ottawa, Canada: Special Surveys Group, Statistics Canada (Four Volumes).

Steinmetz, Carl (1979)

"An empirically tested analysis of victimization risks." Paper presented at the Third International Symposium on Victimology, Münster, Germany, 2-8 September.

Stephan, Egon (1975)

"Die ergebnisse der Stuttgarter opferbefragung unter beruchksicthtigung verglecichbarer Amerikanischer daten." Krimina tatistia 5:201-206.

Stephan, Egon (1976)

Die Stuttgarter Opferbefragung, Wiesbaden, Germany: Bundeskriminalamt

Stephan, Egon (No date)

"Report on two studies of the perception of crime and crime control from the victim's point of view." Freiburg, Germany: Max-Planck-Institute, unpublished manuscript,

Sudman, Seymour, and Norman M. Bradburn (1974)

Response effects in surveys: A review and synthesis. Chicago, Ill.: Aldine.

Traugott, Michael W., and John P. Katosh (1979)

"Response validity in surveys of voting behavior." Public Opinion Quarterly 43 (Fall): 359-377.

Tuchfarber, Alfred, and William R. Klecka (1976)

Random digit dialing: Lowering the cost of victimization surveys. Washington, D.C.: The Police Foundation.

Turner, Anthony G. (1970)

Personal victimization pretest: Evaluation of findings. Washington, D.C.: Statistical Research Center, Law Enforcement Assistance Administration, memorandum, 17 April,

Turner, Anthony G. (1972a)

The San Jose methods test of known crime victims, Washington, D.C.; National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration, U.S. Department of Justice.

Turner, Anthony G. (1972b)

"Methodological issues in the development of the National Crime Survey Panel: Partial findings." Washington, D.C.: National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration, U.S. Department of Justice, memorandum, December.

Turner, Anthony G., and Richard W. Dodge (1972)

"Surveys of personal and organizational victimization." Paper prepared for the Symposium on Studies of Public Experience. Knowledge and Opinion of Crime and Justice, Washington, D.C., March.

Turner, Anthony G. (1977)

"An experiment to compare three interview procedures in the National Crime Survey." Washington, D.C.: Statistical Research Division, U.S. Census Bureau, memorandum, March.

U.S. Census Bureau (1979)

"Survey documentation: National Crime Survey central cities sample, 1974." Washington, D.C.: U.S. Census Bureau, memorandum, bine.

U.S. Department of Justice (1975) Criminal victimization surveys in 13 American cities. Washington, D.C.: National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration.

U.S. Department of Justice (1979a) Criminal victimization in the United States 1976. Washington, D.C.: National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration.

U.S. Department of Justice (1979b) Criminal victimization in the United States: Summary findings of 1977-78 changes in crime and of trends since 1973. Washington, D.C.: National Criminal Justice Information and Statistics Service, Law Enforcement Assistance Administration.

Weiss, Carol M. (1968)

"Validity of welfare mother's interview responses." Public Opinion Quarterly 32 (Winter):622-633.

Wolf, Preben (1976a)

"On individual victims of certain crimes in four Scandinavian countries 1970/74: A comparative study." Paper presented at the Second International Symposium on Victimology, Boston, 5-11 September.

Wolf, Preben (1976b)

Note on some of the methods applied in the Scandinavian victim surveys 1970-74. Kobenhaven: Sociologisk Institut, Kobenhaven Universitet.

Wolfgang, Marvin E. (1978)

National survey of crime severity. Grant Report to LEAA-NCJISS, 9 November.

Woltman, Henry F., John Bushery, and Larry Carstensen (1975)

"Recall bias and telescoping in the National Crime Survey." Washington, D.C.: Statistical Methods Division, U.S. Census Bureau, memorandum, 23 September.

Woltman, Henry F., and Glenn Cadek (1977) "Are memory biases in the National Crime

Survey associated with the characteristics of the criminal incident?" Washington, D.C.: Statistical Methods Division, U.S. Census Bureau, memorandum, 4 April.

Woltman, Henry F., and John M. Bushery (1977a)

"Update of the National Crime Survey Panel bias study." Washington, D.C.: Statistical Methods Division, U.S. Census Bureau, memorandum, 11 July.

Woltman, Henry F., and John M. Bushery (1977b)

"Results of the National Crime Survey maximum personal visit, maximum telephone interview experiment." Washington, D.C.: Statistical Methods Division, U.S. Census Bureau, memorandum, 9 December.

Yost, Linda R., and Richard W. Dodge (1970)

"Household survey of victims of crime: Second pretest-Baltimore, Maryland." Washington, D.C.: U.S. Census Bureau, memorandum, 30 November.

Zimring, Franklin (1972)

"The medium is the message: Firearm caliber as a determinant of death from assault." Journal of Legal Studies 1 (January):97-123.





1

جب المتدابية بالمريبية السقيانية المستسبين

antipi di senara metama darita